



Contents _____

I. Research Projects in 2010

1. Labor Market and Social Policy

0	Measures for Reducing Blind Spots in the Employment Safety Net	4
D	Study on the Work Life and Retirement of the Baby Boom Generation	6
0	Current Conditions and Future Directions of Social Enterprises:	
	with Focus on Employment	8
D	A Study on the Substitutability of Employment among Cohorts	10
D	Changes in the Female Labor Market after the Financial Crisis and Policy Issues	12

2. Human Capital

0	Analysis of the Effect of Green Growth on Jobs:	
	with Focus on Technological Innovation and Climate Change Convention	16
•	An Analysis of Labor Market Mismatch by Educational Attainment and Measures for	
	Improving the Education System: With a Focus on Institutions of Higher Education and	
	Vocational High Schools	18
•	Study on the Entrance of Youth into the Labor Market and Their Establishment	20
0	Analysis of the Effect of Minimum Wage	22

3. Industrial Relations and Workplace Innovation

0	Comparative Study on Changes in Collective Bargaining after the	
	Global Economic Crisis	26
0	Multiple Trade Unions and Full-Time Union Officials:	
	Current Status and Policy Tasks	28
0	Working Time Flexibility and How to Improve It	30
0	A Study on Improving the Employee Representation Plan	33
0	Labor-Management-Government Partnerships for Regional Employment and	
	Measures for Advancement	35

I. Appendix

•	List of Planned Research Projects in 2011	38
•	List of Publications	39
•	Chronology	54

K O R E A L A B O R I N S T I T U T E

I.

Research Projects in 2010

KOREA LABOR INSTITUTE

- 1. Labor Market and Social Policy
 - 2. Human Capital
- 3. Industrial Relations and Workplace Innovation

KLI Annual Report 2010

1. Labor Market and Social Policy

- Measures for Reducing Blind Spots in the Employment Safety Net
- Study on the Work Life and Retirement of the Baby Boom Generation
- Current Conditions and Future Directions of Social Enterprises: with Focus on Employment
- A Study on the Substitutability of Employment among Cohorts
- Changes in the Female Labor Market after the Financial Crisis and Policy Issues

Measures for Reducing Blind Spots in the Employment Safety Net

Jiyeun Chang et al.

his study aims to identify gaps in the employment safety net (ESN) of Korea and present policy options for reducing them. The Korean ESN is based on Employment Insurance (EI), meaning that non-wage workers, including the self-employed, are precluded from the system by design. Moreover, EI coverage for wage workers is only 58.9%. Surveys on micro enterprises (the biggest blind spots in the ESN) reveal employer reluctance to provide EI coverage. One commonly used avoidance option is to register workers in employment types that are exempt from EI when they are in fact wage workers.

A review of the existing EI and its im-

pact at the level of individuals, companies and industries reveals ample grounds for strengthening policy support for the system, for it was found to produce social and economic benefits. To do so, two options are recommended. The first option is to waive the contribution payment for social insurance programs, a policy that is expected to produce clear improvements in job creation for low-wage workers and their transition to wage workers in micro enterprises. Second, a Korea-specific unemployment aid is recommended in order to fill the institutional gaps left uncovered by EI. These aid programs should combine monetary support with strong employment services.

Table of Contents

Chapter 1. Introduction

Chapter 2. Current Status of Blind Spots in the Employment Safety Net

Chapter 3. Employment Insurance and Its Impact on the Period of Unemployment and Poverty

Chapter 4. El and Job Transition/Labor Mobility

Chapter 5. Mechanism for Creating Blind Spots in the ESN

Chapter 6. Wage Preservation Policy and Its Impact on Employment and Wages

Chapter 7. The Second Social Security Net in Japan

Chapter 8. Conclusion: Recommendations for a Stronger ESN

References

☆ KLI Fellows

Jiyeun Chang	-Director, Korea Labor Institute
	-Ph.D. in Sociology, University of Wisconsin-Madison, U.S.A.
Byung-Hee Lee	-Senior Research Fellow, Korea Labor Institute
	-Ph.D. in Economics, Seoul National University, Korea
Soo-Mi Eun	-Research Fellow, Korea Labor Institute
	-Ph.D. in Sociology, Seoul National University, Korea

※ Outside Contributors

ofessor, Kyung Hee University

Study on the Work Life and Retirement of the Baby Boom Generation

Hanam Phang et al.

his study is a follow-up to "Study on the Work Life of Korean Baby Boom Generation (I)," also by Phang, Lee, Woo, Kim, and Kim (2009). The previous study analyzed the socioeconomic characteristics of the baby boomer generation and their current position in the labor market from the perspective that this generation is a single population group. This study also examined the economic burdens of the baby boomer generation in that they have to support their parents, provide for their children's education, and prepare for their own lives after retirement.

Additionally, this study seeks to identify the diversity and differences that exist among the baby boomer generation on a micro level in terms of career paths and results of their work lives as well as in terms of the level of preparation for retirement in human/ material resources such as income, assets, and health.

It can be said that most baby boomers, now in their late 40s to mid 50s, are now at the peak of their work life. Retracing their employment paths will help in examining how socio-economic development and changes over the past 40 years affected the life of the baby boomer generation and how these results are manifested.

In this context, this study analyzes the boomer generation's labor supply structure according to the current status of their assets, consumption tendencies, burden of support, and health. This study also examines the baby boomer generation's level of preparation for life after retirement.

Chapter 6 summarizes the results of the study on the baby boomer generation's work life and their preparations for retirement dealt with in Chapter 2 through Chapter 5. Finally, Chapter 6 presents the implications of the study in terms of labor and social policies as well as topics for future studies.

☼ Table of Contents

Chapter 1. Introduction

Chapter 2. Analysis of the work life of the baby boom generation

Chapter 3. Disparity in education and labor market performance within the baby boom generation

Chapter 4. Assets, consumption, health of the baby boom generation and their labor supply

Chapter 5. Baby boom generation's preparation for life after retirement

Chapter 6. Summary and conclusion

References

Hanam Phang	-Senior Research Fellow, Korea Labor Institute	
	-Ph.D. in Sociology, University of Wisconsin-Madison, U.S.A.	

* Outside Contributors

Sungkyun Lee	-Professor, University of Ulsan
Seokjin Woo	l -Professor, Myongji University
Ki-Hun Kim	-Research Fellow, National Youth Policy Institute
Jikyung Kim	-Research Fellow, National Youth Policy Institute

Current Conditions and Future Directions of Social Enterprises: with Focus on Employment

Charnim Park et al.

obs and employment have become a hot issue in almost every country. Employment concerns are significantly related to increases in the number of chronically unemployed people. Moreover, welfare policies now do not exist by themselves. Rather, they form a strong nexus with employment policies. Previous welfare policies concentrated on providing cash welfare benefits. As the number of benefit recipients who are able to work increased rapidly, however, policies to help them return to the labor market have become more important. In this context, this study examines social enterprises from the perspective of employ-

ment policy. Chapter 2 examines the status of social enterprises within the employment policy framework. Chapter 3 examines the development and characteristics of Korea's welfare system, employment promotion schemes, and social enterprises. Chapter 4 examines the significance of work integration social enterprises as active labor market policy; these enterprises could function as "bridge" jobs for the unemployed in Korea. Chapter 5 discusses the policy direction for the development of work-integration social enterprises in Korea, while Chapter 6 summarizes the study and provides policy directions.

Table of Contents

Chapter 1. Introduction

Chapter 2. Social enterprises and employment

Chapter 3. Development and characteristics of social enterprises in Korea

Chapter 4. Active labor market policies and work-integration social enterprises

Chapter 5. Policy direction for work-integration social enterprises

Chapter 6. Conclusion

References

* KLI Fellows

Charnim Park | -Research Fellow, Korea Labor Institute | -Ph.D. in Social Welfare, Yonsei University, Korea

※ Outside Contributors

Hye-won Kim | -Professor, Korea National University of Education

A Study on the Substitutability of **Employment among Cohorts**

Joyup Ahn

s the 7.1 million baby-boomers born between 1955 and 1963 start to leave the labor market, it is expected that enormous changes will take place in the labor market, such as abrupt skill shortages. To alleviate such negative impacts, many ideas have been discussed about how to make the baby-boomers stay in the labor market longer by extending the retirement age along with initiating a wagepeak system and re-employment after retirement. However, arguments have been raised about measures involved in the "battle of cohorts on jobs" by reducing job opportunities for the young and raising their unemployment. It is a reality that no empirical study exists on the substitutability of employment among cohorts. This study has three objectives: first, it examines the hypothesis of the battle of cohorts on jobs from various aspects; second, it tests the hypothesis that more older workers in the labor market imply higher youth unemployment by using the intergenerational occupation dissimilarity index; finally, an economically rational model is suggested for revising long-term employment contracts with late retirement to make the baby-boomers stay in the labor market at least until they are entitled to a pension. The simple time-series analysis using the data from the Economically Active Population Survey shows no evidence of a statistically significant negative relationship of employment indicators of the old and of the young, which rejects the hypothesis of the battle of cohorts on jobs. The intergenerational occupation dissimilarity index indicates a division of labor based on occupations rather than on competition over occupations, which supports the results from the time-series analysis. Based on the conclusion that there is no battle among cohorts on jobs, it is better to upgrade the mandatory retirement age in accordance with pension entitlement using measures like the wage-peak system for extended retirement, gradual retirement, or

re-employment after retirement.

☆ Table of Contents

Chapter 1. Introduction

Chapter 2. Labor market trends of the young and old

Chapter 3. Testing the hypothesis of the battle of cohorts on jobs-macro analyses

Chapter 4. Testing the hypothesis of the battle of cohorts on jobs-micro analyses

Chapter 5. Occupational dissimilarity index trends

Chapter 6. A protocol for upgrading retirement age

Chapter 7. Conclusions

☆ KLI Fellows

Joyup Ahn | -Senior Research Fellow, Korea Labor Institute | -Ph.D. in Economics, Ohio State University, U.S.A.

Changes in the Female Labor Market after the Financial Crisis and Policy Issues

Jae-Ho Keum et al.

ver since the nation's liberation in 1945, the 1997 Asian financial crisis marked a turning point in Korea's employment history. Understanding the changes in the female labor market in the 12 years from the crisis is important not only for evaluating women's employment policies but also for developing policies for the future. It is from this perspective that the present study reviews changes in major aspects of the women's labor market including employment structure, labor supply, labor demand, occupational distribution, workplace experience and wages.

Chapter 2 focuses on changes in the employment structure from the 1997 crisis to 2009 using data from the Economically Active Population Survey conducted by the National Statistical Office (NSO). Female employment in this period appears to be "half success and half failure." Women's social and economic participation did increase significantly in driving job creation, but it does

far behind men's in terms of the quality of employment. In fact, the employment rate in Korea remains much lower than for other industrialized countries. If the current "jobless growth" continues, the mismatch between demand and supply of female labor can only worsen with possibilities for excluding women from key areas of employment.

Chapter 4 looks at the changes and characteristics of women's economic participation from the perspective of family labor. The Korean Labor and Income Panel Study (KLIPS) data are used to identify the changes in the supply of married women in the labor force as dual-income families continue to increase. As such incomes become the norm, deepening heterogeneity can be found between different groups. The economic participation of married women is closely related to the employment status of their husbands, and participation patterns will vary depending on the wage level of their spouses.

Chapter 5 reviews how certain enterprises manage female human resources and develop policies using the results of a status survey that was conducted on 800 enterprises with over 100 employees. It was found that both the gender ratio and the gender-discriminatory corporate culture have improved considerably in large enterprises with more than 500 employees; however, many smaller companies are mired in old practices. Maternity protection plans remain rare while genderdiscriminatory practices linger to a significant extent in such areas as promotions, wages, positions and training.

Chapter 7 is a review of the extent of occupational gender separation and changes over the years, using data from the NSO's Economically Active Population Survey. In addition, a review of the gender employment ratio and wage gap by occupational category shows that the higher the female ratio in an occupation, the lower the wages. An additional review revealed no actual crowding-out effect.

In Chapter 8, data from the first through the tenth rounds of KLIPS are used to analyze the impact of the number of years of service on women's wages. It was found that in the same workplace, wage level rises with longer years of service. But when women change

their workplace, the years of service in the previous workplace have a positive impact on male wages but not on those of women.

In Chapter 9, changes in the gender wage gap since the financial crisis are analyzed in different contexts. Moreover, the reasons why the gap has remained almost unchanged in recent years are analyzed using 1998-2008 KLIPS data. Since the financial crisis, the wage gap has grown between standard and non-standard workers, and between large companies and SMEs. Consequently, this trend has led to relatively lower wages for women who are in non-standard jobs in SMEs. Although some highly educated women have made strides in technical, professional and managerial jobs, the gender wage gap remains largely unchanged.

As for the basic direction of women's employment policies, this study suggests improving social awareness of the importance of the female workforce through well-organized communication and network-building. In addition, it is recommended that such policies be run more independently and differentiated by group. In particular, given that HR for the female workforce significantly differs by company size, it is also important to differentiate support by company size.

Table of Contents

Chapter 1. Introduction

Chapter 2. Women's Employment Structure: Trends and Changes

Chapter 3. Women's Unemployment Structure: Status and Trends

Chapter 4. Supply of Family/Women's Labor Since the Financial Crisis

Chapter 5. Corporate Management of Female Human Resources

Chapter 6. Labor Market Outcomes of Labor Mobility

Chapter 7. Occupation Segregation: Status and Trends

Chapter 8. Women's Workplace Experience and Wages

Chapter 9. Gender Wage Gap: Status and Policy Tasks

Chapter 10. Policy Tasks and Conclusions

References

Appendix 1. Status Survey on Corporate Management of Female Human Resources

Appendix 2. Policies for Women Since the Financial Crisis

* KLI Fellows

Jae-Ho Keum	-Senior Research Fellow, Korea Labor Institute
	-Ph.D. in Economics, University of Minnesota at Twin Cities, U.S.A.
Jayoung Yoon	-Research Fellow, Korea Labor Institute
	-Ph.D. in Economics, University of Massachusetts-Amherst, U.S.A.

KLI Annual Report 2010

2. Human Capital

- Analysis of the Effect of Green Growth on Jobs: with Focus on Technological Innovation and Climate Change Convention
- An Analysis of Labor Market Mismatch by Educational Attainment and Measures for Improving the Education System: With a Focus on Institutions of Higher Education and Vocational High Schools
- Study on the Entrance of Youth into the Labor Market and Their Establishment
- Analysis of the Effect of Minimum Wage

Analysis of the Effect of Green Growth on Jobs: with Focus on Technological Innovation and Climate Change Convention

Sung-teak Kim et al.

he question of whether green growth actually does promote job creation is extremely important to Korea, for the nation has declared that it will change its economic paradigm to that of a green economy in order to claim leadership in this area in the international community. In countries that have achieved industrialization yet maintain a stable level of economic growth, job creation is a rather difficult policy task. Yet it is one that is still of utmost priority. Usually, green growth is thought of as growth realized through technological innovation. Such innovation leads to economic development that in turn brings about the creation of employment opportunities. The idea that green growth creates more employment has been criticized, however, from the following standpoints. First, can we really claim that technological innovation in the green sector promotes job creation when employment is actually decreasing in the manufacturing industry in which

technological innovations for this growth are centered? Second, will green growth not lead to a decrease in employment due to the heavier cost burden levied on businesses in the form of new regulations and environmental taxes along with the introduction of carbon emissions trading and other changes in the economic paradigm associated with green growth? And third, will employment actually decrease in industries that face restructuring pressures as the paradigm shifts towards green growth?

This report consists of six chapters elaborating on the aforementioned concerns about job creation as related to green growth. Chapter 2 discusses the relationship among green growth, green jobs and job creation and notes limitations to the analysis of these relationships. Chapter 3 views the example of Germany, a country that has pursued green growth for quite some time, and the German experience of creating jobs in this sector. The most notable positions in this

country have evolved through programs involving renewable energy and building retrofitting projects. Thus, this study seeks to analyze how education and training related to the process of green growth was utilized in Germany. Chapter 4 analyzes previous studies to examine the influence of technological innovation on economic growth and job creation. In so doing, our objective is to see whether green-technology innovation, an important key to green growth, does in fact ensure the creation of new jobs. Chapter 5 attempts to analyze the impact of the Climate Change Convention on jobs. The notion of green growth grew out of the need for countries to fulfill their carbon emission reduction obligations as agreed upon at this convention. The virtuous cycle of green growth leads to increased investment, the expansion of new industries and rosy prospects for more jobs. On the other hand, the Climate Change Convention can lead to a vicious cycle of increased costs, decreased production and a smaller number of jobs in Korean businesses. The interaction of these two somewhat contradictory forces must be analyzed to foresee the influence of this interaction on job creation and destruction, and to identify other factors that can inspire job creation within the context of the interaction of these contradictory forces. Throughout this analysis, we seek to examine which approaches may be helpful for climate change and green growth policies to act as positive factors within Korea's labor market. Chapter 6 provides a final summary of the previous chapters and sets forth an overview of the policy implications of our analysis.

* Table of Contents

Chapter 1. Introduction

Chapter 2. Green Growth and Green Jobs

Chapter 3. Green Growth and Job Creation: The Example of Germany

Chapter 4. Technological Innovation and Job Creation

Chapter 5. The Impact of the Climate Change Convention on the Labor Market

Chapter 6. Summary and Conclusion

References

* KLI Fellows

Sung-teak Kim	-Acting President, Korea Labor Institute
	-Ph.D. in Economics, Brown University, U.S.A.

☼ Outside Contributors

Dong-Soon Im	-Professor, Dong-Eui University	

An Analysis of Labor Market Mismatch by Educational Attainment and Measures for Improving the Education System: With a Focus on Institutions of Higher Education and Vocational High Schools

Se-Um Kim et al.

his study focuses on making changes to the formal schooling system as a long-term and fundamental solution to the mismatch in workforce supply-demand, the key culprit behind falling youth employment. The aim of the first section is to raise awareness about the need for university restructuring. In addition, future policy direction implications are presented in a quantitative assessment of the mismatch in Korea's labor market for university graduates. Also presented is an analysis of the cause for the slow progress of university restructuring despite the need for this change. The second section raises a call for fostering specialized high schools such as Meister schools by providing a quantitative assessment of the mismatch in the labor market for vocational high school graduates, as well as policy alternatives regarding methods of support. The final section is based on the finding that the workforce supply-demand mismatch observed in the Korean labor market is ultimately caused by the trend of excessive emphasis on enrollment in post secondary schools. This cultural characteristic, unique to Korea, will be assessed quantitatively in the final section, which includes the social costs of the prevalence of such a trend and a review of the appropriateness of the current policy goal of reducing shadow education expenditures. Also examined will be the historical background of this trend, and based on the findings, a step-by-step policy approach will be presented for changing the culture in the long term. This study aims to establish awareness of the necessity of improving Korea's formal schooling system to resolve the mismatch in workforce supply-demand through empirical analysis. These disparities are regarded as an urgent issue in terms of national employment strategy. At the same time, insights will be provided on the policy measures required to facilitate institutional improvement. Finally, suggestions will be made for future studies.

☼ Table of Contents

Chapter 1. Introduction

Chapter 2. Analysis of the Status Survey on Corporate HR Managers

Chapter 3. Labor Market Mismatch for University Graduates and Restructuring of the University System

Chapter 4. Career Guidance in Vocational High Schools and How to Improve the Employment Rate

Chapter 5. "University-first" Culture: Historical Background and Social Costs

Chapter 6. Conclusion

References

* KLI Fellows

Se-Um Kim	-Research Fellow, Korea Labor Institute
	-Ph.D. in Economics, Brown University, U.S.A.

X Outside Contributors

Jin Yeong Kim	-Professor, Konkuk University
Young Jun Cho	-Research Professor, Kyujanggak Institute for Korean Studies, Seoul National University

Korea Labor Institute 19

Study on the Entrance of Youth into the Labor Market and Their Establishment

Jaeryang Nam et al.

his study reviews the main characteristics of the youth labor market to deepen understanding of the youths and conducts an analysis that will help to develop appropriate policies. Specifically, this study raises the issue of the rapid increase in young interns, identifies some characteristic facts including increases in non-standard youth workers and evaluates the role and results of such internship programs to analyze their implications. Because the youth labor market is highly sensitive to business cycles and displays dynamic activities, this study also undertakes dynamic analysis on how the youth labor market reacts to cyclical changes. It also examines the rising educational level in Korea (evidenced by the high university enrollment rate),

which is leading to qualitative mismatches due to higher expectations as well as quantitative mismatches in labor supply and demand.

Because of the difficulties experienced by youth in entering and staying in the labor market, some may call for policies to protect youth livelihoods. As a result, this study builds a model to analyze the impact on youth employment of direct/indirect subsidies that may be made available for the youths under the minimum wage system, Earned Income Tax Credit or the Basic Livelihood Protection Program. Such an evaluation will help provide a more objective and broader understanding of youth and help develop more plausible solutions to their problems.

☼ Table of Contents

Chapter 1. Introduction

Chapter 2. Main Characteristics of the Youth Labor Market

Chapter 3. Increase in Youth Internship Programs and Labor Market Entry and Settlement

Chapter 4. Employment Types of Youth Workers and Labor Market Entry and Settlement

Chapter 5. Cyclical Changes and Youth Unemployment: Analysis of the Patterns of Youth Unemployment and Job-seeking

Chapter 6. Causes of Long-term Non-employment of University Graduates and Policy Implications

Chapter 7. Worker Support Policy and Youth Unemployment: Comparison among EITC, Minimum Wage System and Wage Subsidy

Chapter 8. Conclusion

References

* KLI Fellows

Jaeryang Nam	-Research Fellow, Korea Labor Institute
	-Ph.D. in Economics, Seoul National University, Korea

※ Outside Contributors

Cheol-In Lee	-Professor, Seoul National University
Young Jun Chun	-Professor, Hanynag University
Seokjin Woo	-Professor, Myongji University

Analysis of the Effect of Minimum Wage

Jin-Ho Jeong et al.

he minimum wage system, long considered to be the most effective policy measure to support the working poor, is in need of a more comprehensive analysis for enhancing its effectiveness. Its coverage has continued to expand since 1988, but in spite of a history of more than 20 years, research has not been conducted on the different effects of the minimum wage. Nevertheless, the minimum wage is still receiving great attention as a wage adjustment method, which is more significant than collective bargaining for SMEs, non-standard workers and migrant workers. Likewise, the Earned Income Tax Credit (EITC) was introduced in 2009 to support the working-poor, and its policy effectiveness has not been analyzed.

Against this backdrop, this study aims to find ways to improve the minimum wage system based on an empirical analysis of its effectiveness. In particular, this study examines the effects of the minimum wage on employment; wage and earning distribution; and income distribution by means of a detailed categorization (e.g., skill level, income level, etc.) to identify policy targets and to offer specific policy measures for these targets.

☼ Table of Contents

Chapter 1. Introduction

Chapter 2. The Development of the Minimum Wage

Chapter 3. The Effects of the Minimum Wage on Employment

Chapter 4. The Effects of the Minimum Wage on Wages and Earnings Distribution

Chapter 5. The Effects of the Minimum Wage on Income Distribution

Chapter 6. An Evaluation of Policy Alternatives: The Minimum Wage, EITC, and Other Alternatives

References

* KLI Fellows

Jin-Ho Jeong	-Director-General, Korea Labor Institute	
	-Ph.D. in Economics, Seoul National University, Korea	
Jaeryang Nam	-Research Fellow, Korea Labor Institute	
	-Ph.D. in Economics, Seoul National University, Korea	

※ Outside Contributors

Juyoung Kim	-Research Fellow, Korea Institute for Industrial Economics and Trade
Young Jun Chun	-Professor, Hanyang University

Korea Labor Institute 23

3

KLI Annual Report 2010

3. Industrial Relations and Workplace Innovation

- Comparative Study on Changes in Collective Bargaining after the Global Economic Crisis
- Multiple Trade Unions and Full-Time Union Officials: Current Status and Policy Tasks
- · Working Time Flexibility and How to Improve It
- A Study on Improving the Employee Representation Plan
- Labor-Management-Government Partnerships for Regional Employment and Measures for Advancement

Comparative Study on Changes in Collective Bargaining after the Global Economic Crisis

Changwon Lee et al.

his study compares how labor, management and government in Korea and other industrialized countries have conducted concession bargaining or work-sharing in response to the global financial crisis, to see how it is associated with general concession bargaining and labor-management relations. It also reviews the different ways of work-sharing undertaken in Korea to identify relevant policy implications.

The patterns of concession bargaining since the global financial crisis can be largely defined as follows. First, concessions in wage and employee benefits have increased, along with work-sharing by reducing work hours, which until then had been implemented on a very limited basis. Second, pronounced cooperation has been growing among labor, management and government in response to the financial crisis. The government's role in particular is gaining more importance. Third, designing different solutions in accordance

with circumstances in individual companies has become more important than ever. Such trends are becoming stronger.

One pattern of response unique to Korea as opposed to other industrialized countries is that wage cuts, wage freezes or temporary leave have been used as important means to ride out the financial storm. Such measures were made available because of Korea's relatively healthy fiscal position and business environment along with the intent of government and individual companies to avoid restructuring, which has had a strong impact on their options.

During the financial crisis, an agreement was reached on implementing work-sharing through a government-labor-management-civil society pact, but the actual implementation and results were highly limited compared to what has been done in other industrialized countries. The reason can be found in Korea's reliance on long work hours in companies for their workers.

☼ Table of Contents

Chapter 1. Introduction

Chapter 2. Concession Bargaining in Europe after the Financial Crisis

Chapter 3. Concession Bargaining in Korea after the Financial Crisis

Chapter 4. Work-sharing in Industrialized Countries

Chapter 5. Work-sharing in Korea

Chapter 6. Conclusion

References

Changwon Lee	-Senior Research Fellow, Korea Labor Institute	
	-Ph.D. in Sociology, Chicago University, U.S.A.	

X Outside Contributors

Sang-Mok Kim	-Researcher, Posco Research Insitute
Young-Ho Lee	-Former Director for Planning and Management Team, Korea Workplace Innovation Center
Heon-Hyeok Ha	-Former Director for Workplace Innovation Support Team, Korea Workplace Innovation Center

Multiple Trade Unions and Full-Time Union Officials: Current Status and Policy Tasks

Sung-Hee Lee et al.

n January 1, 2010, a bill introducing multiple trade unions in an enterprise and prohibiting payment to full-time union officials was passed by the National Assembly; therefore, major changes will occur in industrial relations in Korea. Workers will be allowed to form more than one trade union in their workplaces, and companies will be prohibited from paying full-time union officials as the time-off system is introduced. Such institutional change is likely to bring about significant changes in a variety of issues in industrial relations including the establishment of trade unions, trade union activities, patterns of collective bargaining, and the characteristics of industrial relations.

Against such a backdrop, this study aims to provide an outlook for the changes in industrial relations following the approval of multiple trade unions and the introduction of the time-off system. Another objective of the study is to provide ways to ensure the

soft landing of these new systems, to find productive industrial relations practices, and to present policy alternatives for building a new framework in industrial relations.

To analyze how the establishment of multiple trade unions will affect industrial relations, employers and trade union members were surveyed. Based on the results, this study seeks to analyze the number of additional trade unions that will be formed, how the number of union members will change, how the nature of the collective bargaining system and industrial relations will be transformed, and what key issues will become prominent between labor and management when the bill takes effect in July 2011.

Based on these empirical analyses, this study provides an outlook for the changes in industrial relations following the introduction of multiple trade unions and the time-off system and suggests policy tasks for establishing productive labor-management relationships.

☼ Table of Contents

Chapter 1. Purpose and structure of the study

Chapter 2. Outlook survey for industrial relations in the era of multiple trade unions

Chapter 3. Time-off system

Chapter 4. Industrial relations after the legislative revisions on multiple trade unions and full-time officials

Chapter 5. Changes concerning multiple trade unions and full-time officials and policy tasks

References

* KLI Fellows

Sung-Hee Lee	-Research Fellow, Korea Labor Institute	
	-Ph.D. in Business Administration, Korea University, Korea	
Jung-Han Kim	-Research Fellow, Korea Labor Institute	
	-Ph.D. in Business Administration, Sogang University, Korea	
Seong Jae Cho	-Research Fellow, Korea Labor Institute	
	-Ph.D. in Economics, Seoul National University, Korea	

**** Outside Contributors**

Yung-Woo Son	-Lecturer, Dongguk University
Hyunji Kwon	-Lecturer, Kings College London

Korea Labor Institute 29

Working Time Flexibility and How to Improve It

Kiu-Sik Bae et al.

his study looks at the characteristics, degree and background of working time flexibility in Korea, while exploring future directions for such flexibility. An extremely small number of Korean studies exist on working time flexibility, and Korean enterprises have not been able to utilize or explore this topic fully. This study approaches this topic through a comprehensive perspective based upon a critical analysis of previous studies dealing with working time flexibility from the viewpoints of changing external demand; family and employment institutions; work-life balance; life stages; and the sharing and creation of jobs. While based on a review of previous literature, also provided in this study are analyses and comparisons of working hour statistics from Korea and abroad, domestic case studies and examples of working time flexibility in the Netherlands, Germany and the UK.

Under the system of long working hours

established during Korea's industrialization process, working time flexibility has usually referred to the extension of working hours by adjusting overtime hours for a relatively small number of workers in accordance with changes in external demand. Working hours in Korea have decreased considerably over the past two decades with reductions to statutory working hours that have led to a decrease in both standard working hours and overtime hours. This reduction has brought considerable quantitative change to the system of long working hours in Korea.

Still, the system of long working hours in Korea has managed to retain its elemental characteristics as it adjusts to external change. Flexible working hours and other work practices involving flexibility have been used in very limited cases in Korean businesses, and the ratio of part-time work is also very low due to the prevalent model of long working hours for workers who are predominantly single-income men. Even in

workplaces involving occupations not affected by white collar exemptions that apply to clerical, technical, sales and professional positions (e.g., the banking industry), increasing the number of extra working hours by simply including overtime pay for a few hours each week in the basic wage is still prevalent. Also, the Special Provisions as to Working and Recess Hours as stated in Article 59 of the Labor Standards Act provide for exemptions from working time regulations for the transport industry and many other industries, thus providing life support to the system of long working hours in Korea. In addition, the automobile, shipbuilding, machinery and some other manufacturing industries also reinforce long working hours amidst labor and management collusion. The sustainability of long working hours is weakening, however, due to their failure to meet social demands for working hour distribution with regard to changes in the external environment; changing demographics; changes to the structure of the labor market; family and employment institutions; work-life balance; and life stages.

Countries such as the Netherlands, Germany and the UK are each utilizing unique models of working time flexibility according to the characteristics of their labor markets. The Netherlands has expanded the utilization of varied and individualized regular working hours to enhance working time flexibility in a model of "1.5 person per household employment" realized through a higher female em-

ployment rate and the sharing of social jobs. Germany has heightened working time flexibility by introducing a working time account system for workers to deposit or withdraw working time according to fluctuations in demand from businesses or as needed by the workers. Meanwhile, the UK has increased working time flexibility by expanding parttime work and by introducing several flexible types of work arrangements.

This study espouses the need to redesign the working hour system in Korea to enhance flexibility while still meeting demands related to changes in the external environment, changes to the structure of the labor market, changing demographics, work-life balance and different life stages. To this end, labor, management and the government need to work together to reduce working hours through statutory regulation, reforming the system of long working hours and establishing a regular part-time work model for the public sector. In addition, efforts should be made to introduce various flexible working time practices, transparently manage the working hours of clerical, technical, service and sales workers, and grant existing fulltime workers the right to move to part-time work or other flexible work arrangements. Being exploratory in nature, this study aims to research working time flexibility in Korea systematically and to analyze various cases to identify exactly what aspects of this multifaceted working time flexibility are being utilized and to examine why other aspects are limited in their use. Finally, the task of furthering theoretical discussions on this topic is left to future studies.

* Table of Contents

Chapter 1. Introduction

Chapter 2. Theoretical Discussion on Flexible Working Hours

Chapter 3. Working Hour Regimes in Korea and Working Time Flexibility

Chapter 4. Utilization of Flexible Working Hours in Korea

Chapter 5. Flexible Working Hours in Three European Countries

Chapter 6. Comparison of Flexible Working Hours in Three Countries and Implications

Chapter 7. Summary and Challenges for Flexible Working Hours

References

★ KLI Fellows

Kiu-Sik Bae	-Director-General, Korea Labor Institute
	-Ph.D. in Industrial and Business Studies, University of Warwick, U.K.

**** Outside Contributors**

Sang-min Lee	-Professor, Hanyang University
Hyunji Kwon	-Lecturer, Kings College London

A Study on Improving the **Employee Representation Plan**

Hoon Kim et al.

his study begins from the awareness that the collective determination system for working conditions has been stretched to its limits, and therefore some measures should be undertaken. The union organization rate has been on a declining trend, diminishing the scope of the collective bargaining system. Although the transition to industrial unions is underway, the existing paradigm of company-level unions, where wage and other working conditions are determined at individual companies, remains largely unchanged. On the other hand, the increased flexibility of the labor market has diversified employment types, creating different worker groups (regular/part-time/fixed-term/ dispatch/in-house subcontracting) with different interests within the same company or workplace. In addition, because the HR system is adjusted to place more emphasis

on performance, even workers in the same employment type have different working conditions (e.g., regular jobs).

As workers become more and more individualized in the labor market and even in the same workplace, questions arise on how effective the existing unions are led by regular workers in large companies in coordinating or representing the diverse interests of the worker groups. If the unions at the center of the collective determination system are not fully effective in defining working conditions and setting norms, an alternative may be found in the "employee representation system." It is within this context that this study reviews issues regarding the current employee representation system and analyzes how it can be adjusted or re-built for the better. This study also presents some principles for related legislation.

Table of Contents

Chapter 1. Introduction

Chapter 2. Need to Re-build the Employment Representation System

Chapter 3. Issues of the Employee Representation System under the Existing Labor Relations Act

Chapter 4. The Labor-Management Council in Non-unionized Workplaces

Chapter 5. Employee Participation and Business Performance

Chapter 6. The Employee Representation System in Japan: Legislation and Implications

Chapter 7. Conclusion: How to Re-build the Employee Representation System

★ KLI Fellows

Hoon Kim	-Senior Research Fellow, Korea Labor Institute -Ph.D. in Industrial Sociology, Keio University, Japan
Jung-woo Kim	-Senior Researcher, Korea Labor Institute -Ph.D. in Economics, Inha University, Korea

**** Outside Contributors**

Jong-hee Park	-Professor, Korea University		
---------------	------------------------------	--	--

Labor-Management-Government Partnerships for Regional Employment and Measures for Advancement

Myung-Sook Jun et al.

o enable analysis of the current status of regional labor-management-government partnerships, this study will first examine theoretical discussions for conceptualizing such phenomena. Moreover, this study seeks to analyze the recent trend of increased participation of the stakeholders in a region: labor, employers, civic groups, and local governments in policy-making and execution processes through the concepts/theoretical tools of "Meso Corporatism", "local pact", and the "associational model". By utilizing such theoretical tools, this study will analyze case examples of various regional labor-management-government partnerships that have recently been formed in Korea. In particular, topics will be analyzed including types of labor-management-government partnerships, governance, method of organization management, relationship with the central government, and problems.

Case examples involving regional labor-

management-government partnerships in Korea can be categorized according to broad and narrow definitions: 1) In a broad sense, regional labor-management-government partnerships refer to various policies and projects in which stake holders in the region participate—labor, employers, and local government—to strengthen partnerships. 2) In a narrow sense, regional labormanagement-government partnerships refer to specific policies and projects, in particular the increasingly common "regional labormanagement-civic groups-government partnership consultative bodies" along with projects conducted through such bodies.

In the case of the former, regional labormanagement-government partnerships are realized by the stakeholders of a region participating in policy execution and forming a "local pact" to undertake projects. In the case of the latter, regional labor-management-government partnerships are realized through the "associational model" in which

the stakeholders of a region participate in policy-making and execution through an officialized governance system.

This study will analyze the current status

and present measures for advancement for both definitions of labor-management-government partnerships.

* Table of Contents

Chapter 1. Introduction

Chapter 2. Development of the Tripartite Partnership for Regional Employment in Korea

Chapter 3. Case Study 1: Policy-Making Partnership

Chapter 4. Case Study 2: Project-Based Partnership

Chapter 5. Case Study of Japan

Chapter 6. Conclusion

References

☆ KLI Fellows

Myung-Sook Jun	-Former Research Fellow, Korea Labor Institute	
	-Professor, Chonnam University	
	-Ph.D. in Industrial Relations, University of Wisconsin Madison, U.S.A.	
Hong Geun Chang	-Research Fellow, Korea Labor Institute	
	-Ph.D. in Sociology, Seoul National University, Korea	

※ Outside Contributors

Yong-bo Shim	-Research Fellow, Korea Labor Foundation
Sang-hoon Lim	-Professor, Hanyang University

Appendix

KOREA LABOR INSTITUTE

List of Planned Research Projects in 2011

List of Publications

Chronology

List of Planned Research Projects in 2011

Research on Labor Market and Social Policy

Marriage migrant families and labor market integration

Worker household consumption and savings and the determination of labor supply

Study on the fixed-term contractual labor market

Compatibility between labor mobility and the social security system

A study on job creations of care services sector

Economic policies to improve job creation effect of growth

Employment-oriented welfare strategies

🔆 Research on Human Capital

Study on wage system innovation in step with the aging population

A study on changes in job preferences by generation

Research on Industrial Relations and Workplace Innovation

Study on measures for more active conciliation in individual labor disputes

Developing a model to measure costs for collective bargaining and empirical analysis

The social enterprise and human resource strategies

Study on a model for workplace innovation in the services industry

Managing working hours: shift from long to shorter working hours

List of Publications

Periodicals

Monthly Labor Review (Monthly) International Labor Brief (Monthly) Quarterly Journal of Labor Policy (Quarterly)

* Publications in English

Beyond Flexibility: Roadmaps for Korean Labor Policy	August 29, 2008
Globalization and Changes in Employment Conditions in Asia and the Pacific	December 30, 2007
Longitudinal Analysis of Inter-Industry Wage Differentials in the Korean Labor Market	June 25, 2007
Employment Insurance in Korea: The First Ten Years	August 23, 2006
The Transformation of Industrial Relations in Large-size Enterprises in Korea	March 29, 2006
Recent Development in Employment Relations and Labor Market in Korea	December 31, 2005
A Comparative Study on Labor Migration Management in Selected Countries	November 15, 2004
Labor in Korea: 1987-2002	June 18, 2004
Disadvantaged Groups in the Labour Market and Employment Policy	March 11, 2003
Active Labor Market Policies and Unemployment Insurance in Selected Countries	December 23, 2002
Labor in Korea	December 12, 2002
North Korean Labor and Its Prospects	May 1, 2001
The Profile of Korean Human Assets: Labor Statistics 2001	May 1, 2001
Labor Market Reform in Korea: Policy Options for the Future	March 1, 2001
The Employment Insurance System in Korea	November 8, 1999

Labor Market and Employment Policy

Improving the Job Creation Capability of the Services Industry by Upgrading the Employment Structure [III] December 30, 2010 with a Focus on the Culture and Tourism Sectors Analysis of Labor Market Mismatch by Educational Attainment and Measures for Improving the Education System with a Focus on Institutions of Higher Education and Vocational High Schools The Employment and Training Policies of the United States Non-Market Time and Children's Human Capital Formation Study on Reentering the Labor Market Entrepreneurship and Labor Study on Reentering the Labor Market Entrepreneurship and Labor Study on Do Creation through Green Growth June 10, 2010 Economic Crisis and Employment Study on the Work Life of the Baby-Boomer Generation in Korea May 28, 2010 Development of the Employment Intermediary Industry and Job Creation I How to Strengthen the Job Creation Capacity of the Service Industry For the Advancement of the Employment Structure A Study on the Creation and Destruction of Jobs Wage Differences in Korea July 12, 2009 Wational Pension: Impact on Labor Supply and Savings June 12, 2009 Wational Pension: Impact on Labor Supply and Savings June 12, 2009 Wage Study on Freelancers: Focusing on the Film and IT Industries An Analysis of the Self-Employment Strategies of the Service Industry A Study on Freelancers: Focusing on the Film and IT Industries An Analysis of the Self-Employment Strategies of the Service Industry A Study on Freelancers: Focusing on the Film and IT Industries An Analysis of the Self-Employment Strategies of the Service Industry A Study on Freelancers: Focusing on the Film and IT Industries An Analysis of the Self-Employment Strategies of Morea A Study on Korea's Future Paradigm for Social and Economic Policies [II] A Study on Freelancers: Focusing on the Film and IT Industries Analysis and Assessment of Employment Impact) Coctober 31, 2008 Policy Measures to Promote Efficienc	Labor Market and Employment Policy	
Change Convention An Analysis of Labor Market Mismatch by Educational Attainment and Measures for Improving the Education System with a Focus on Institutions of Higher Education and Vocational High Schools The Employment and Training Policies of the United States Non-Market Time and Children's Human Capital Formation Study on Reentering the Labor Market Entrepreneurship and Labor Study on Reentering the Labor Market Entrepreneurship and Labor Study on Job Creation through Green Growth Loune 10, 2010 Economic Crisis and Employment Study on the Work Life of the Baby-Boomer Generation in Korea Development of the Employment Intermediary Industry and Job Creation I May 14, 2010 How to Strengthen the Job Creation Capacity of the Service Industry For the Advancement of the Employment Structure A Study on the Creation and Destruction of Jobs Wage Differences in Korea Analysis of the Self-Employed Labor Market [II] Empirical Study on Job Creation in the Third Sector National Pension: Impact on Labor Supply and Savings History of Working Lives of Mid-to-Senior Aged Workers Competitiveness and Employment Strategies of the Service Industry A Study on Freelancers: Focusing on the Film and IT Industries An Analysis of the Self-Employment Strategies of the Service Industry A Study on Freelancers: Focusing on the Film and IT Industries An Analysis of the Self-Employment Strategies in Korea A Study on Korea's Future Paradigm for Social and Economic Policies [II] Analyzing the Impact of Industrial Policies on Job Creation 12007 Research Series of Analysis and Assessment of Employment Impact! Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008 Current Status and Development Strategies of Regional Employment and Human Resource Development	Improving the Job Creation Capability of the Services Industry by Upgrading the Employment Structure [II] with a Focus on the Culture and Tourism Sectors	December 30, 2010
Education System with a Focus on Institutions of Higher Education and Vocational High Schools The Employment and Training Policies of the United States Non-Market Time and Children's Human Capital Formation Study on Reentering the Labor Market June 30, 2010 Entrepreneurship and Labor Study on Job Creation through Green Growth June 10, 2010 Study on Job Creation through Green Growth Economic Crisis and Employment Study on the Work Life of the Baby-Boomer Generation in Korea May 28, 2010 Study on the Work Life of the Baby-Boomer Generation in Korea May 25, 2010 Development of the Employment Intermediary Industry and Job Creation I How to Strengthen the Job Creation Capacity of the Service Industry For the Advancement of the Employment Structure A Study on the Creation and Destruction of Jobs August 14, 2009 Wage Differences in Korea Analysis of the Self-Employed Labor Market [III] July 15, 2009 Empirical Study on Job Creation in the Third Sector June 30, 2009 National Pension: Impact on Labor Supply and Savings History of Working Lives of Mid-to-Senior Aged Workers Competitiveness and Employment Strategies of the Service Industry A Study on Freelancers: Focusing on the Film and IT Industries An Analysis of the Self-Employed Labor Market Employment Insurance and Activation Strategies in Korea A Study on Korea's Future Paradigm for Social and Economic Policies [II] Analyzing the Impact of Industrial Policies on Job Creation (2007 Research Series of Analysis and Assessment of Employment Impact) Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008 Current Status and Development Strategies of Regional Employment and Human Resource Development		December 30, 2010
Non-Market Time and Children's Human Capital Formation Study on Reentering the Labor Market Entrepreneurship and Labor Study on Job Creation through Green Growth June 10, 2010 Economic Crisis and Employment May 28, 2010 Study on the Work Life of the Baby-Boomer Generation in Korea May 25, 2010 Development of the Employment Intermediary Industry and Job Creation 1 How to Strengthen the Job Creation Capacity of the Service Industry For the Advancement of the Employment Structure A Study on the Creation and Destruction of Jobs August 14, 2009 Wage Differences in Korea July 24, 2009 Analysis of the Self-Employed Labor Market [III] Empirical Study on Job Creation in the Third Sector June 30, 2009 Mational Pension: Impact on Labor Supply and Savings History of Working Lives of Mid-to-Senior Aged Workers Competitiveness and Employment Strategies of the Service Industry A Study on Freelancers: Focusing on the Film and IT Industries Analysis of the Self-Employed Labor Market May 22, 2009 A Study on Freelancers: Focusing on the Film and IT Industries Analysis of the Self-Employed Labor Market May 22, 2009 A Study on Korea's Future Paradigm for Social and Economic Policies [II] A Study of Employment Strategies for Job Creation: 2008 Employment Strategies Analysis (2007 Research Series of Analysis and Assessment of Employment Impact) October 31, 2008 Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008		December 20, 2010
Study on Reentering the Labor Market Entrepreneurship and Labor Study on Job Creation through Green Growth June 10, 2010 Economic Crisis and Employment May 28, 2010 Study on the Work Life of the Baby-Boomer Generation in Korea May 25, 2010 Development of the Employment Intermediary Industry and Job Creation I How to Strengthen the Job Creation Capacity of the Service Industry For the Advancement of the Employment Structure A Study on the Creation and Destruction of Jobs August 14, 2009 Wage Differences in Korea Analysis of the Self-Employed Labor Market [II] Empirical Study on Job Creation in the Third Sector June 30, 2009 National Pension: Impact on Labor Supply and Savings History of Working Lives of Mid-to-Senior Aged Workers Competitiveness and Employment Strategies of the Service Industry A Study on Freelancers: Focusing on the Film and IT Industries May 29, 2009 A Study on Freelancers: Focusing on the Film and IT Industries A Study on Korea's Future Paradigm for Social and Economic Policies [III] April 15, 2009 A Study of Employment Strategies for Job Creation: 2008 Employment Strategies Analysis [2007 Research Series of Analysis and Assessment of Employment Impact] Analyzing the Impact of Industrial Policies on Job Creation [2007 Research Series of Analysis and Assessment of Employment Impact] October 31, 2008 Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008	The Employment and Training Policies of the United States	July 28, 2010
Entrepreneurship and Labor Study on Job Creation through Green Growth Economic Crisis and Employment May 28, 2010 Study on the Work Life of the Baby-Boomer Generation in Korea May 25, 2010 Development of the Employment Intermediary Industry and Job Creation I How to Strengthen the Job Creation Capacity of the Service Industry For the Advancement of the Employment Structure A Study on the Creation and Destruction of Jobs August 14, 2009 Wage Differences in Korea July 24, 2009 Wage Differences in Korea July 15, 2009 Empirical Study on Job Creation in the Third Sector June 30, 2009 National Pension: Impact on Labor Supply and Savings June 12, 2009 History of Working Lives of Mid-to-Senior Aged Workers Competitiveness and Employment Strategies of the Service Industry A Study on Freelancers: Focusing on the Film and IT Industries May 29, 2009 A Analysis of the Self-Employed Labor Market May 22, 2009 A Study on Korea's Future Paradigm for Social and Economic Policies [III) February 27, 2009 A Study of Employment Strategies for Job Creation: 2008 Employment Strategies Analysis [2007 Research Series of Analysis and Assessment of Employment Impact] Analyzing the Impact of Industrial Policies on Job Creation [2007 Research Series of Analysis and Assessment of Employment Impact] October 31, 2008 Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008	Non-Market Time and Children's Human Capital Formation	July 16, 2010
Study on Job Creation through Green Growth Economic Crisis and Employment May 28, 2010 Study on the Work Life of the Baby-Boomer Generation in Korea May 25, 2010 Development of the Employment Intermediary Industry and Job Creation I How to Strengthen the Job Creation Capacity of the Service Industry For the Advancement of the Employment Structure A Study on the Creation and Destruction of Jobs August 14, 2009 Wage Differences in Korea Analysis of the Self-Employed Labor Market [II] Empirical Study on Job Creation in the Third Sector National Pension: Impact on Labor Supply and Savings History of Working Lives of Mid-to-Senior Aged Workers Competitiveness and Employment Strategies of the Service Industry A Study on Freelancers: Focusing on the Film and IT Industries An Analysis of the Self-Employed Labor Market Employment Insurance and Activation Strategies in Korea A Study on Korea's Future Paradigm for Social and Economic Policies [II] Astudy of Employment Strategies for Job Creation: 2008 Employment Strategies Analysis and Assessment of Employment Impact) Analyzing the Impact of Industrial Policies on Job Creation Projects [2007 Research Series of Analysis and Assessment of Employment Impact] Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008	Study on Reentering the Labor Market	June 30, 2010
Economic Crisis and Employment Study on the Work Life of the Baby-Boomer Generation in Korea May 28, 2010 Development of the Employment Intermediary Industry and Job Creation I How to Strengthen the Job Creation Capacity of the Service Industry For the Advancement of the Employment Structure A Study on the Creation and Destruction of Jobs August 14, 2009 Wage Differences in Korea Analysis of the Self-Employed Labor Market [II] Empirical Study on Job Creation in the Third Sector National Pension: Impact on Labor Supply and Savings History of Working Lives of Mid-to-Senior Aged Workers Competitiveness and Employment Strategies of the Service Industry A Study on Freelancers: Focusing on the Film and IT Industries Analysis of the Self-Employed Labor Market Employment Insurance and Activation Strategies in Korea A Study on Korea's Future Paradigm for Social and Economic Policies [II] A Study of Employment Strategies for Job Creation: 2008 Employment Strategies Analysis (2007 Research Series of Analysis and Assessment of Employment Impact) Analyzing the Impact of Industrial Policies on Job Creation Projects [2007 Research Series of Analysis and Assessment of Employment Impact) Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008	Entrepreneurship and Labor	June 18, 2010
Study on the Work Life of the Baby-Boomer Generation in Korea Development of the Employment Intermediary Industry and Job Creation I How to Strengthen the Job Creation Capacity of the Service Industry For the Advancement of the Employment Structure A Study on the Creation and Destruction of Jobs August 14, 2009 Wage Differences in Korea July 24, 2009 Analysis of the Self-Employed Labor Market [II] Empirical Study on Job Creation in the Third Sector June 30, 2009 National Pension: Impact on Labor Supply and Savings June 12, 2009 History of Working Lives of Mid-to-Senior Aged Workers June 5, 2009 Competitiveness and Employment Strategies of the Service Industry A Study on Freelancers: Focusing on the Film and IT Industries May 29, 2009 A Study on Korea's Future Paradigm for Social and Economic Policies [II] A Study of Employment Strategies for Job Creation: 2008 Employment Strategies Analysis and Assessment of Employment Impact) Analyzing the Impact of Industrial Policies on Job Creation (2007 Research Series of Analysis and Assessment of Employment Impact) Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008 October 31, 2008 Current Status and Development Strategies of Regional Employment and Human Resource Development	Study on Job Creation through Green Growth	June 10, 2010
Development of the Employment Intermediary Industry and Job Creation I How to Strengthen the Job Creation Capacity of the Service Industry For the Advancement of the Employment Structure A Study on the Creation and Destruction of Jobs August 14, 2009 Wage Differences in Korea July 24, 2009 Analysis of the Self-Employed Labor Market [II] Empirical Study on Job Creation in the Third Sector National Pension: Impact on Labor Supply and Savings June 30, 2009 History of Working Lives of Mid-to-Senior Aged Workers Competitiveness and Employment Strategies of the Service Industry A Study on Freelancers: Focusing on the Film and IT Industries May 25, 2009 An Analysis of the Self-Employed Labor Market Employment Insurance and Activation Strategies in Korea A Study on Korea's Future Paradigm for Social and Economic Policies [II] A Study of Employment Strategies for Job Creation: 2008 Employment Strategies Analysis [2007 Research Series of Analysis and Assessment of Employment Impact] Analyzing the Impact of Industrial Policies on Job Creation [2007 Research Series of Analysis and Assessment of Employment Impact] October 31, 2008 Current Status and Development October 31, 2008 Current Status and Development	Economic Crisis and Employment	May 28, 2010
How to Strengthen the Job Creation Capacity of the Service Industry For the Advancement of the Employment Structure A Study on the Creation and Destruction of Jobs Wage Differences in Korea July 24, 2009 Analysis of the Self-Employed Labor Market (III) July 15, 2009 Empirical Study on Job Creation in the Third Sector June 30, 2009 National Pension: Impact on Labor Supply and Savings History of Working Lives of Mid-to-Senior Aged Workers June 5, 2009 Competitiveness and Employment Strategies of the Service Industry A Study on Freelancers: Focusing on the Film and IT Industries An Analysis of the Self-Employed Labor Market May 22, 2009 Employment Insurance and Activation Strategies in Korea A Study on Korea's Future Paradigm for Social and Economic Policies (III) A Study of Employment Strategies for Job Creation: 2008 Employment Strategies Analysis (2007 Research Series of Analysis and Assessment of Employment Impact) Analyzing the Impact of Industrial Policies on Job Creation Projects (2007 Research Series of Analysis and Assessment of Employment Impact) Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008	Study on the Work Life of the Baby-Boomer Generation in Korea	May 25, 2010
Employment Structure A Study on the Creation and Destruction of Jobs August 14, 2009 Wage Differences in Korea July 24, 2009 Analysis of the Self-Employed Labor Market (III) July 15, 2009 Empirical Study on Job Creation in the Third Sector National Pension: Impact on Labor Supply and Savings June 12, 2009 History of Working Lives of Mid-to-Senior Aged Workers June 5, 2009 Competitiveness and Employment Strategies of the Service Industry A Study on Freelancers: Focusing on the Film and IT Industries May 29, 2009 An Analysis of the Self-Employed Labor Market May 22, 2009 Employment Insurance and Activation Strategies in Korea A Study on Korea's Future Paradigm for Social and Economic Policies (III) A Study of Employment Strategies for Job Creation: 2008 Employment Strategies Analysis (2007 Research Series of Analysis and Assessment of Employment Impact) Analyzing the Impact of Industrial Policies on Job Creation (2007 Research Series of Analysis and Assessment of Employment Impact) Policy Measures to Promote Efficiency of Job Creation Projects (2007 Research Series of Analysis and Assessment of Employment Impact) Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008	Development of the Employment Intermediary Industry and Job Creation I	May 14, 2010
Wage Differences in Korea July 24, 2009 Analysis of the Self-Employed Labor Market [II] July 15, 2009 Empirical Study on Job Creation in the Third Sector June 30, 2009 National Pension: Impact on Labor Supply and Savings History of Working Lives of Mid-to-Senior Aged Workers June 5, 2009 Competitiveness and Employment Strategies of the Service Industry A Study on Freelancers: Focusing on the Film and IT Industries May 29, 2009 A Study on Freelancers: Focusing on the Film and IT Industries May 22, 2009 Employment Insurance and Activation Strategies in Korea A Study on Korea's Future Paradigm for Social and Economic Policies [II] A Study of Employment Strategies for Job Creation: 2008 Employment Strategies Analysis [2007 Research Series of Analysis and Assessment of Employment Impact] Analyzing the Impact of Industrial Policies on Job Creation [2007 Research Series of Analysis and Assessment of Employment Impact] Policy Measures to Promote Efficiency of Job Creation Projects (2007 Research Series of Analysis and Assessment of Employment Impact) Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008		February 19, 2010
Analysis of the Self-Employed Labor Market (II) Empirical Study on Job Creation in the Third Sector National Pension: Impact on Labor Supply and Savings History of Working Lives of Mid-to-Senior Aged Workers June 12, 2009 Competitiveness and Employment Strategies of the Service Industry A Study on Freelancers: Focusing on the Film and IT Industries May 29, 2009 An Analysis of the Self-Employed Labor Market May 22, 2009 Employment Insurance and Activation Strategies in Korea A Study on Korea's Future Paradigm for Social and Economic Policies (III) A Study of Employment Strategies for Job Creation: 2008 Employment Strategies Analysis (2007 Research Series of Analysis and Assessment of Employment Impact) Analyzing the Impact of Industrial Policies on Job Creation (2007 Research Series of Analysis and Assessment of Employment Impact) Policy Measures to Promote Efficiency of Job Creation Projects (2007 Research Series of Analysis and Assessment of Employment Impact) Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008	A Study on the Creation and Destruction of Jobs	August 14, 2009
Empirical Study on Job Creation in the Third Sector National Pension: Impact on Labor Supply and Savings History of Working Lives of Mid-to-Senior Aged Workers June 12, 2009 Competitiveness and Employment Strategies of the Service Industry A Study on Freelancers: Focusing on the Film and IT Industries May 25, 2009 An Analysis of the Self-Employed Labor Market Employment Insurance and Activation Strategies in Korea A Study on Korea's Future Paradigm for Social and Economic Policies (III) A Study of Employment Strategies for Job Creation: 2008 Employment Strategies Analysis (2007 Research Series of Analysis and Assessment of Employment Impact) Analyzing the Impact of Industrial Policies on Job Creation (2007 Research Series of Analysis and Assessment of Employment Impact) Policy Measures to Promote Efficiency of Job Creation Projects (2007 Research Series of Analysis and Assessment of Employment Impact) Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008	Wage Differences in Korea	July 24, 2009
National Pension: Impact on Labor Supply and Savings June 12, 2009 History of Working Lives of Mid-to-Senior Aged Workers June 5, 2009 Competitiveness and Employment Strategies of the Service Industry A Study on Freelancers: Focusing on the Film and IT Industries May 29, 2009 An Analysis of the Self-Employed Labor Market Employment Insurance and Activation Strategies in Korea A Study on Korea's Future Paradigm for Social and Economic Policies (III) A Study of Employment Strategies for Job Creation: 2008 Employment Strategies Analysis (2007 Research Series of Analysis and Assessment of Employment Impact) Analyzing the Impact of Industrial Policies on Job Creation (2007 Research Series of Analysis and Assessment of Employment Impact) Policy Measures to Promote Efficiency of Job Creation Projects (2007 Research Series of Analysis and Assessment of Employment Impact) Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008	Analysis of the Self-Employed Labor Market (II)	July 15, 2009
History of Working Lives of Mid-to-Senior Aged Workers Competitiveness and Employment Strategies of the Service Industry A Study on Freelancers: Focusing on the Film and IT Industries May 29, 2009 An Analysis of the Self-Employed Labor Market May 22, 2009 Employment Insurance and Activation Strategies in Korea April 15, 2009 A Study on Korea's Future Paradigm for Social and Economic Policies (III) February 27, 2009 A Study of Employment Strategies for Job Creation: 2008 Employment Strategies Analysis (2007 Research Series of Analysis and Assessment of Employment Impact) Analyzing the Impact of Industrial Policies on Job Creation (2007 Research Series of Analysis and Assessment of Employment Impact) Policy Measures to Promote Efficiency of Job Creation Projects (2007 Research Series of Analysis and Assessment of Employment Impact) Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008	Empirical Study on Job Creation in the Third Sector	June 30, 2009
Competitiveness and Employment Strategies of the Service Industry A Study on Freelancers: Focusing on the Film and IT Industries May 25, 2009 An Analysis of the Self-Employed Labor Market May 22, 2009 Employment Insurance and Activation Strategies in Korea April 15, 2009 A Study on Korea's Future Paradigm for Social and Economic Policies (II) February 27, 2009 A Study of Employment Strategies for Job Creation: 2008 Employment Strategies Analysis (2007 Research Series of Analysis and Assessment of Employment Impact) Analyzing the Impact of Industrial Policies on Job Creation (2007 Research Series of Analysis and Assessment of Employment Impact) Policy Measures to Promote Efficiency of Job Creation Projects (2007 Research Series of Analysis and Assessment of Employment Impact) Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008	National Pension: Impact on Labor Supply and Savings	June 12, 2009
A Study on Freelancers: Focusing on the Film and IT Industries An Analysis of the Self-Employed Labor Market Employment Insurance and Activation Strategies in Korea April 15, 2009 A Study on Korea's Future Paradigm for Social and Economic Policies (III) A Study of Employment Strategies for Job Creation: 2008 Employment Strategies Analysis (2007 Research Series of Analysis and Assessment of Employment Impact) Analyzing the Impact of Industrial Policies on Job Creation (2007 Research Series of Analysis and Assessment of Employment Impact) Policy Measures to Promote Efficiency of Job Creation Projects (2007 Research Series of Analysis and Assessment of Employment Impact) Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008	History of Working Lives of Mid-to-Senior Aged Workers	June 5, 2009
An Analysis of the Self-Employed Labor Market Employment Insurance and Activation Strategies in Korea April 15, 2009 A Study on Korea's Future Paradigm for Social and Economic Policies (III) A Study of Employment Strategies for Job Creation: 2008 Employment Strategies Analysis (2007 Research Series of Analysis and Assessment of Employment Impact) Analyzing the Impact of Industrial Policies on Job Creation (2007 Research Series of Analysis and Assessment of Employment Impact) Policy Measures to Promote Efficiency of Job Creation Projects (2007 Research Series of Analysis and Assessment of Employment Impact) Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008	Competitiveness and Employment Strategies of the Service Industry	May 29, 2009
Employment Insurance and Activation Strategies in Korea A Study on Korea's Future Paradigm for Social and Economic Policies (II) A Study of Employment Strategies for Job Creation: 2008 Employment Strategies Analysis (2007 Research Series of Analysis and Assessment of Employment Impact) Analyzing the Impact of Industrial Policies on Job Creation (2007 Research Series of Analysis and Assessment of Employment Impact) Policy Measures to Promote Efficiency of Job Creation Projects (2007 Research Series of Analysis and Assessment of Employment Impact) Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008	A Study on Freelancers: Focusing on the Film and IT Industries	May 25, 2009
A Study on Korea's Future Paradigm for Social and Economic Policies (II) A Study of Employment Strategies for Job Creation: 2008 Employment Strategies Analysis (2007 Research Series of Analysis and Assessment of Employment Impact) Analyzing the Impact of Industrial Policies on Job Creation (2007 Research Series of Analysis and Assessment of Employment Impact) Policy Measures to Promote Efficiency of Job Creation Projects (2007 Research Series of Analysis and Assessment of Employment Impact) Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008	An Analysis of the Self-Employed Labor Market	May 22, 2009
A Study of Employment Strategies for Job Creation: 2008 Employment Strategies Analysis (2007 Research Series of Analysis and Assessment of Employment Impact) Analyzing the Impact of Industrial Policies on Job Creation (2007 Research Series of Analysis and Assessment of Employment Impact) Policy Measures to Promote Efficiency of Job Creation Projects (2007 Research Series of Analysis and Assessment of Employment Impact) Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008	Employment Insurance and Activation Strategies in Korea	April 15, 2009
Series of Analysis and Assessment of Employment Impact) Analyzing the Impact of Industrial Policies on Job Creation (2007 Research Series of Analysis and Assessment of Employment Impact) Policy Measures to Promote Efficiency of Job Creation Projects (2007 Research Series of Analysis and Assessment of Employment Impact) Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008	A Study on Korea's Future Paradigm for Social and Economic Policies (II)	February 27, 2009
Assessment of Employment Impact) Policy Measures to Promote Efficiency of Job Creation Projects (2007 Research Series of Analysis and Assessment of Employment Impact) Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008		October 31, 2008
Assessment of Employment Impact) Current Status and Development Strategies of Regional Employment and Human Resource Development October 31, 2008		October 31, 2008
Current Status and Development Strategies of Regional Employment and Human Resource Development (2007 Research Series of Analysis and Assessment of Employment Impact)		October 31, 2008
	Current Status and Development Strategies of Regional Employment and Human Resource Development [2007 Research Series of Analysis and Assessment of Employment Impact]	October 31, 2008

Study of Employment Impact Analysis, and Assessment Methodologies (2007 Research Series of Analysis and Assessment of Employment Impact)	October 31, 2008
WPS 2005 Workplaces and Labor in Korea: Preliminary Analysis Report on Workplace Panel	October 20, 2008
Child Education and Labor Supply of Married Women	September 16, 2008
A Study on the Dynamic Characteristics of the Labor Market	July 30, 2008
Workforce Structure of Professional Performing Artists and Policy Assistance: A Focus on Performers in Plays and Musicals	May 30, 2008
An Analysis on the Impact of Job Mobility on the Labor Market	May 30, 2008
Job Transition Process of Involuntarily Unemployed Workers	May 30, 2008
A Statistical Study on Two Decades of Labor	May 30, 2008
A Study on Regional Labor Market (IV): Establishing an Exemplary Regional Partnership—How to Set Up the Necessary Institutions	May 28, 2008
International Comparative Study on Industry-Level Labor Flexibility: Focusing on Corporate Case Comparisons	May 20, 2008
Structural Changes in the Labor Market and Shifts in Employment	April 15, 2008
Global Comparison of the Labor Market and Employment of Older Workers	April 11, 2008
Employment Flexibility and Non-Regular Employment (Study Series on the Labor Issues of the Past 20 Years since 1987)	April 4, 2008
Changes in the Structure of Workers' Skills since 1987 (Study Series on the Labor Issues of the Past 20 Years since 1987)	April 4, 2008
Low-Earning Labor Market in Korea	February 29, 2008
Status of Employment Service for Underprivileged Workers and Service Improvement Measures	August 28, 2007
Endogenous Local Development Strategies and Local Partnership: With a Focus on a Garment Industry Cluster	August 10, 2007
$Study on \ Regional \ Labor \ Markets \ III: Status \ of \ the \ Labor \ Market \ of \ Upper \ Level \ Local \ Government$	August 10, 2007
Reshaping of Work-Incentive System in North Korea	June 30, 2004
A Fair Globalization: Creating Opportunities for All	June 15, 2004
A Study on Policies for Women Employment in Some OECD Members: The UK, Canada, Sweden, and Denmark	July 10, 2007
A Study on the Labor Supply Effect of Earned Income Tax	June 29, 2007
An Analysis of the Labor Market for Migrant Workers and Directions to Improve the Management System	June 29, 2007
Employment Opportunities in the Rural Labor Market: Implications From Statistical Analysis and Case studies	June 15, 2007
Labor and Discrimination (II): Awareness and Practice	June 8, 2007
Comparison of Worker Participation between Korea and Japan	June 5, 2007
An Empirical Study on the Correlation between Workers' Health Conditions and Labor Supply	June 5, 2007
Structural Adjustment of Korean Economy and Service Employment	May 30, 2007

A Study on Employment Adjustment of Korean Firms	May 18, 2007
Skill Formation of Core Technical Manpower I : The Changes in Korea's Skill Structure and the Exploration of Core Technical Manpower	May 15, 2007
Economic Analysis on the Labor Market Polarization in Korea	April 30, 2007
Conceptualization and Measurement of Quality of Employment	March 30, 2007
A Study on the Assistance Plan for Workers Affected by the Free Trade Agreement	February 28, 2007
Discrimination in the Labor Market and Affirmative Action I	December 30, 2006
Status of the Self-Employed and Policy Tasks	December 6, 2006
An Analysis of Unemployment of Youth with Education Levels Not Exceeding High School Graduation and Policy Tasks	November 30, 2006
Employment Relations and Labor Issues at Call Centers	September 12, 2006
A Study on the Causes of Low Birthrate for Workers and Women-Friendly Labor Market Policies	September 11, 2006
Evaluation of the 2004 Job Creation Program	June 23, 2006
Labor Market Policies in Industrialized Nations at \$10,000 to 20,000 Per Capita GDP	May 8, 2006
Employment Relations in Small and Medium-Sized Manufacturers	April 28, 2006
The Current Status of Anti-Poverty Training Policy and Tasks	April 21, 2006
Employment Policy to Counter the Korea-Japan FTA	April 21, 2006
Changes in the Corporate Investment Environment and the Labor Market	April 12, 2006
Pay Structure and Performance-Based Pay in Korea	April 10, 2006
Trade Liberalization and Its Impact on the Labor Market	April 3, 2006
Study on Lifelong Learning and the Labor Market	March 24, 2006
Study on Cultural Industry Professionals	March 20, 2006
Regional Labor Market Analysis and Policy Recommendations: An Analysis of Labor Market and Local Councils of Bucheon and Policy Recommendations	March 17, 2006
Division of Labor and Employment Relations in the Manufacturing Industry of Northeast Asia (II)	March 10, 2006
Labor Market Polarization and Policy Tasks	March 7, 2006
Macro-Economic Policies: Impact on Employment	March 7, 2006
Promoting Private Job Placement Service	February 28, 2006
Employment Stability in an Age of Flexibility: Evidence From industrialized Countries	February 28, 2006
Korean Model for Industrial Relations (II)	February 17, 2006
Labor Demand and Supply: Mid-/Long-term Prospects	November 30, 2005
Situation of Employment Instability and Employment Policy Tasks	September 15, 2005
A Study on Strategies for Jobless Growth [I]	August 31, 2005

A Study on Strategies for Jobless Growth (II)	August 31, 2005
An Analysis of the Labor Market of the Expert and Technical Foreign Workforce	August 29, 2005
Population Aging and the Labor Market	August 26, 2005
Labor Mobility and Skill Development	July 15, 2005
North Korean Defectors: Employment and Policy Tasks	July 11, 2005
Employment Policies of Korea	June 1, 2005
Demand and Supply of IT Experts in 2003: Analysis and Forecast	May 6, 2005
Wages and the Labor Market in Korea	April 20, 2005
Determining the Impact of ICT on Decent Work	April 15, 2005
Labor Market Analysis of Industrial Trainees for Overseas-Invested Firms in Korea	April 6, 2005
Vitalizing the Regional Labor Market: Model Country Studies and Policy Recommendations	March 31, 2005
A Study on the Working Poor in Korea	March 30, 2005
Education and the Labor Market in Korea	March 30, 2005
Free Movement of Human Resources Through the Korea-China-Japan FTA	December 30, 200
An Analysis of the Turnover of the Insured Using the Employment Insurance DB	August 30, 2004
Utilization of Child-Care Leave and Future Policy Directions	August 30, 2004
Unemployment Benefits in the United Kingdom	August 30, 2004
Project Evaluation of Job Skill Development Program in Korea	August 27, 2004
A Comparative Study on Labor Migration Management	August 16, 2004
Employment Promotion Training Program and Employment Prospects	August 13, 2004
Labor Market Analysis of Low-Skilled Migrant Workers in Korea	August 4, 2004
Maternity Protection for Female Workers and Policy Agenda	June 25, 2004
Labor Market Policies in the Aging Era II	May 28, 2004
Toward a "Flexicurity" in the Korean Labor Market: What Has to Be Done?	May 25, 2004
Vocational Qualification and the Labor Market in Korea	May 15, 2004
A Study on the Evolution and Causes of Wage Differentials by Educational Attainment in Korea	May 12, 2004
Working-Hour Reduction and Employment Policies	May 10, 2004
Changes in Employment Structure of the Knowledge-Based Service Sector	May 10, 2004
Basic Report on the 5th Annual Korean Labor and Income Panel Study	May 10, 2004
Labor in the Financial Industry Before and After the Recent Economic Crisis	May 10, 2004
The Effect of Information and Communication Technology on Workplace Participation: From the Perspective of Decent Work	April 20, 2004

Part-Time Work as Women's Alternative Employment Option	March 5, 2004
Demand for Manpower in the Information Technology Sector	September 9, 2003
Efficiency Assessment of the Employment Stabilization Program	July 31, 2003
Labor Market and Employment Policy in an Aging Era	June 13, 2003
Labor Market for Aged Workers	May 28, 2003
The Current Status of the Private Human Resource Service Industry in Korea and Policy Challenges	May 9, 2003
Changes in the Internal Labor Market in Corporations	April 25, 2003
Vision and Direction of the Vocational Qualification System in Korea	April 21, 2003
Non-Standard Workers and Policy Implications (III)	April 18, 2003
Labor Market Analysis and Policy Recommendations for Secondary-School Graduates	March 25, 2003
The Analysis of Local Labor Markets	March 21, 2003
The Digital Economy and Job Creation	March 10, 2003
The Labor Market for Women Workers and Challenges	June 13, 2002
Longer-Term Forecast of Demand for and Supply of Labor: 2002–2010	April 30, 2002
Evaluation of the Temporary Work System (TWS) and Proposals for Its Improvement	April 30, 2002
Wage Flexibility and Unemployment	April 20, 2002
Legal and Institutional Revision for Reduction of Working Hours	April 15, 2002
Employment Adjustment of the United Kingdom; Focusing on Redundancy	April 9, 2002
Labor Mobility and Unemployment	April 8, 2002
Transition from School to the Labor Market in Korea	April 4, 2002
Information and Communications Technologies (ICTs) and the Labor Market	March 29, 2002
Non-Standard Employment and Policy Suggestions	March 30, 2002
Employment Adjustment and the Outplacement Service Program (OSP)	January 12, 2002
Socioeconomic Effects of Working-Hour Reduction	September 10, 2001

• Industrial Relations and Human Resource Management

Industrial Relations in Korean Businesses Operating in Vietnam	December 15, 2010
Pay Determination in the UK Civil Service	November 30, 2010
A Comparative Research of On-the-Job Training in Korea and Japan	September 30, 2010
Local Human Resource Development Strategies through Corporate Networks	August 20, 2010

Study on Balancing Management Rights and Bargaining Rights in the Collective Bargaining System of the Public Sector	August 10, 2010
Legal Principles for Determining Working Conditions with a Focus on Written Agreements with Worker Representatives	June 30, 2010
Research on Vitalizing Works Councils for Government Employees in the Era of Government Employee Labor Unions	June 18, 2010
Categorization of Workplaces and Innovation Results	June 15, 2010
Theories and Practices of Industrial Bargaining: A Comparison by Industry and Country	October 30, 2009
Foreign Direct Investment and Labor Policy Issues: Comparative Research Covering the UK, Ireland, Singapore, and China	October 20, 2009
Regional Human Resources Development and Employment Governance	October 20, 2009
Labor-Management Relations in Foreign-Invested Companies in Korea	September 15, 2009
Policies to Improve Public Organizations and Labor-Management Relations	September 10, 2009
Women's Labor Dispute Settlement Mechanism and Improvement	September 10, 2009
A Study on the Labor Relations Commission and Improvement of the Labor Dispute Settlement System	August 25, 2009
Skill Development System and Labor-Management Relations: Cases of Korea, Japan, and Germany	August 10, 2009
Employment Relations in the Distribution and Service Industries: With a Focus on Restaurants, Hotels, Department Stores, and Discount Department Stores	June 30, 2009
Privatization Issues and Labor-Management Relations	June 30, 2009
Comparison of Staffing-Quota Management in the Public Sector of Korea and Japan	June 10, 2009
A Study on the Response to Social Responsibility Policies: With a Focus on ISO 26000	May 29, 2009
Research on Building a Social Consensus on Socioeconomic Development Model (NRCS Joint Research Series)	April 30, 2008
An Analysis of Employment Relations in the Public Sector Using the KLI Workplace Panel Survey (WPS)	March 10, 2009
Policy Proposals to Improve Labor Relations for Government Employees	December 12, 2008
Organization of Labor Unions in the Public Sector and Case Studies of Collective Bargaining	December 10, 2008
Technological Innovation and Human Resource Management	November 20, 2008
Leadership Crisis in Korea's Labor Movement: Discussions on the Political Branding that Union Leaders are Self-Centered	October 24, 2008
Changes in Employment Relations in the Financial Service Sector : Focusing on Non-Standard Workers Providing Services to Customers	September 5, 2008
Korea's Employers' Organizations: Korea Employers' Federation, Metal Industries Employers' Council, Health & Medical Industries Employers' Council	August 27, 2008
A Study on the Organization of the Government Employees Union	July 30, 2008
Changes in the Korean Industrial Relations System and Non-Regular Workers (I): Comparing Korea, Japan, and the United States	July 30, 2008
A Study on the Korean Labor Union Organization: Focusing on the Analysis of the Number of Union Members (Organization Rate)	June 10, 2008

An Analysis of Employment Regulations (II)	June 5, 2008
A Study on European Industry-Level Collective Bargaining and Collective Agreements	May 28, 2008
Corporate Social Responsibility and Labor: 2007 CSR Labor Forum Papers	May 16, 2008
Changes in the Korean Economy and Labor System	May 9, 2008
Trade Unions and the Labor Movement Since 1987: Changes in the Korean Industrial Relations System and Future Prospects	April 30, 2008
Industry-Level Labor Relations: A Feasible Future?	April 30, 2008
Two Decades of the Labor System since 1987: Evaluation and Future Prospects (Study Series on the Labor Issues of the Past 20 Years since 1987: Summary Report)	April 28, 2008
Public Opinion Survey Report on Industrial Relations (Study Series on the Labor Issues of the Past 20 Years since 1987)	April 10, 2008
A Study on the Change in Industrial Relations and Actual Business Conditions for Korean Companies in China	March 31, 2008
A study on the Change of Fairness Perception of Korean Employees	March 10, 2008
Union Democracy and Worker Organizations: Focusing on Worker Organizations at Hyundai Motors, KIA Motors, and Daewoo Motors	February 28, 2008
Industrial Relations in the Public Sector: Case Study of the Metropolitan Region (Seoul, Incheon, and Gyeonggi)	December 12, 2007
Technological Innovation and Human Resources Development	August 6, 2007
Labor Division Structure and Employment Relations in the Northeast Asian Manufacturing Industry (III)	June 29, 2007
New Public Management and the Labor Movement of Public Employees	June 29, 2007
Working Conditions of Non-Regular Workers: A Focus on Warranty-Service Repairmen and Bus Drivers for Private Educational Institutions	June 28, 2007
Technology Innovation and Labor Relations	May 31, 2007
Employment Relations at Non-Union Firms: With a Special Regard to Labor-Management Councils	May 30, 2007
The Formation of Labor Relations in Local Government Bodies	May 7, 2007
Atypical Employment and Transformation of Industrial Relations in Korea (I)	April 23, 2007
The Spread of Corporate Social Responsibility and Implications for Industrial Relations	December 30, 2006
An Analysis of Prerequisites for Social Dialogue	July 14, 2006
A Study on Labor Relations under the Multiple Unions Environment	April 7, 2006
How to Stabilize Industrial Relations in Korea: With a Focus on the Major Industries	March 21, 2006
Labor Relations of Public Employees With a Focus on Collective Bargaining	March 15, 2006
Labor Movement Ideologies of Korea	December 30, 2005
Characteristics and Policy Tasks of the Public Workers' Union in Korea: As Revealed Through Background Analysis and Survey	October 20, 2005
Training and Education in the Construction Industry: How to Promote Labor-Management-Government Participation	September 28, 2005

A Study on the Korean Industrial Relations System	August 26, 2005
The Structure of Labor Demand in Korea	July 5, 2005
International and Comparative Employment Relations	June 30, 2005
The Structure and Characteristics of Labor Strikes in Korea	June 20, 2005
Labor's Participation in Management: Status and Challenges	June 10, 2005
The Structure of Labor Division and Employment Relations in the Northeast Asian Manufacturing Industry (I)	May 20, 2005
Pay Structure and Performance-Based Pay in Korea	March 28, 2005
Public-Sector Restructuring, Changes in Industrial Relations, and Stabilization Measures	July 30, 2004
Collective Bargaining at the Industry Level and Policy Implications	June 30, 2004
Japanese Social Consultation Against Globalization	June 25, 2004
Firm-Level Labor-Management Agreement for Job Creation	June 11, 2004
Labor Market and Industrial Relations in China	May 12, 2004
Subcontracting Structure and Stratification of Employment Relations in the Korean Automotive Industry	April 26, 2004
Flexibility and Human Resources Management	April 19, 2004
Social Pacts in Selected Countries	February 10, 2004
Labor Activism, Collective Bargaining, and Dispute Settlement in the U.S. Public Sector	December 26, 2003
International Comparison of Industrial Relations Models	October 13, 2003
Labor in Korea: 1987-2002	September 18, 2003
New Labor Relations in the Knowledge-Based Industries	July 18, 2003
Policy Concertation and Social Partnership in Western Europe	May 12, 2003
Non-Standard Work and Industrial Relations	April 22, 2003
Industrial Relations and HRM of Korean Firms Invested in China	March 28, 2003
The Future of Labor and a New Paradigm	March 20, 2003
Evaluation and Future Development of the Korean Tripartite Commission	February 28, 2003
A Study on Labor-Management Participatory Vocational Competency Development and the Labor Learning Fund	December 20, 2002
Antecedents and Consequences of Employee Participation	November 29, 2002
A Study on Labor and Management Relations in Korea: Analysis on Preliminary Panel Survey on the Workplace	May 20, 2002
A Study on Developing Rehabilitation Programs for Pneumoconiosis Workers: Medical Treatment and Poverty Alleviation	April 30, 2002
Analysis on Collective Agreement (III)	April 25, 2002
A Study on Labor and Management Relations in Korea: Basic Statistics	April 4, 2002
Human Resources Development of Enterprises and Policy Challenges	March 29, 2002

• Labor Laws and Regulations

2008 Review of Labor Jurisprudences	August 20, 2009
In-House Subcontracting and Labor Laws	April 30, 2009
Labor Law Issues in 2007 (II): A Collection of Papers from the Forums on Labor Law and Legal Economics in the Second Half of 2007 $$	November 20, 2008
OECD Monitoring of Korean Labor Laws	July 25, 2008
A Study on Transitional Labor Relations	June 20, 2008
Legal Principles Determining Working Conditions at Non-Unionized Workplaces	April 18, 2008
Labor Law Issues in 2007 [I]: A collection of Papers from the Forums on Labor Law and Legal Economics in the First Half of 2007	February 22, 2008
Expanded Application of the Labor Standard Act: Impact and Effective Approach	October 31, 2007
Labor Law Issues in 2006: A Collection of Papers from the Forums on Labor Law and Legal Economics in 2006	June 21, 2007
Labor and Discrimination (I): Antidiscrimination Systems of Some Advanced Countries	May 30, 2007
Challenges for Labor Laws to Achieve Equal Treatment for Workers	May 15, 2007
Corporate Group and the Labor Law : With a Focus on the Labor-Management Council and the Collective Bargaining System	March 30, 2007
Wage Structure Reform: Challenge for Labor Laws	May 23, 2006
Manpower Management at the Gaeseong Industrial District and Analyses of Labor Laws	May 15, 2006
Social Pacts of the Weak	March 24, 2006
A Study on the Unfair Labor Practice Remedial System	August 31, 2005
Analysis of the Labor Union Constitution (II)	August 31, 2005
A Legalistic Study on Diversification of Trade Union Structure	October 29, 2004
An International Comparison of Equal Employment Laws	August 31, 2004
Changes in the Definition of Workers and the Imposition of Pertinent Laws	July 16, 2004
A Comparative Study on the Linkage between Special Economic Zones and Global Labor Standards	May 28, 2004
A Study on Actual Conditions of North Korean Workers and Their Utilization	May 7, 2004
Labor-Related Laws and Institutions in Japan	December 30, 2003
ILO Labor Legislation Guidelines	August 18, 2003
A Labor Law-Based Approach to Recognizing Special Employment Workers as Laborers	April 30, 2003
Effective Remedy for Unfair Dismissal	April 22, 2003
Proposals for the Direction of Legislation of Basic Labor Rights for Civil Servants	April 9, 2003
A Legal Study on Standards for Selecting Workers for Dismissal for Economic Reasons	March 28, 2003
International Labor Standards and Industrial Relations in Korea	March 15, 2003

An Outlook on Changes in Labor Law Systems of North Korea: Based on an Analysis of China	December 20, 2002
Changes in Collective Labor Relations Laws of Advanced Countries and Policy Challenges	August 30, 2002
Wage Bargaining in 2000 and its Policy Implications	December 15, 2001
Diversification of Employment Types and Institutional Revision	September 25, 2001

• Workers' Welfare and Social Insurance

Study on the Lifetime Income Redistribution of Social Security Systems June 30, 2010 Job Security and Family Structure June 10, 2010 The Working Poor in Korea: Analysis and Policy International Comparison of Make Work Pay Policies and Implications for the Korean Welfare System A Study on the 3rd Sector in Health and Welfare June 30, 2009
The Working Poor in Korea: Analysis and Policy May 7, 2010 International Comparison of Make Work Pay Policies and Implications for the Korean Welfare System February 19, 2010
International Comparison of Make Work Pay Policies and Implications for the Korean Welfare System February 19, 2010
A Study on the 3rd Sector in Health and Welfare June 30, 2009
A Study on Gradual Retirement and the Partial Pension System April 10, 2009
A Korean Model for Social Policies: Linking Employment and Welfare October 15, 2008
Social Insurance and the Role of Labor Unions: Comparing Korea, Germany, and Japan September 26, 2008
Health and Labor Supply of Middle- and Old-Aged People: An Empirical Approach July 21, 2008
A Study on How to Apply Employment Insurance to Economically Dependent Workers June 27, 2008
Unemployment Benefits Paid to Day Laborers and Challenges for Institutional Improvement: Focusing on the Construction Industry June 25, 2008
Compatibility of the Welfare System and the Labor System: Dynamics of the Productive Welfare System since 1987 (Study Series on the Labor Issues of the Past 20 Years since 1987) April 4, 2008
Technical Skill Promotion Programs for National Competitiveness October 22, 2007
Industrial Safety and Health and Worker Participation: Comparison of Korea and Germany October 15, 2007
Premium Rates of Industrial Accident Insurance by Business in 2007 May 31, 2007
An Economic Analysis on the Social Safety Net: With a Focus on Labor Market Effects April 27, 2007
Conditions, Issues, and Policy Tasks of Special Types of Occupations and Relevant Industries November 30, 2006
Unification of Social Insurance Application and Collection Systems and Future Tasks October 30, 2006
Calculation of the 2006 Industrial Accident Insurance Premium Rate by Job Category April 28, 2006
A Study on How to Create Social Service Jobs February 17, 2006
Population Aging and Employment Policies of Korea November 30, 2005
A Study on Advancing the Industrial Accident Compensation Insurance System (Rehabilitation & Welfare) October 31, 2005
Comparison of Workers Compensation Insurance: With a Focus on Therapy September 30, 2005

A Study on the Workers Compensation System for Workers of Atypical Types of Employment	September 12, 200
Simulating Workplace Safety Policies in Korea	August 31, 2005
The Effects of Providing the Voluntarily Separated with Job-Seeking Benefits in Korea	August 16, 2005
Rehabilitation Programs for Industrial Accidents in Major Countries	July 29, 2005
Youth Unemployment and Targeted Policy Recommendations	July 20, 2005
A Study on the Labor Mobility of Injured Workers	July 15, 2005
Types and Legal Status of Pseudoworkers (Special-Type Employees): The German Case	May 20, 2005
Effects of the Reemployment Bonus System in Korea	April 15, 2005
Reconciling Work and Family Life in Selected OECD Countries	March 20, 2005
The History of Employment Insurance in its First 10 Years	December 1, 2005
Industrial Accident Compensation Insurance Forum (2004, Winter)	December 31, 2004
Four Decades of Industrial Accident Compensation Insurance	November 4, 2004
Foreign Industrial Accident Insurance Schemes	August 30, 2004
Development of Social Adjustment Program Model and Approaches for Workers Injured in Industrial Accidents	August 27, 2004
Industrial Accident Compensation Insurance Forum (2004, Summer)	July 31, 2004
The French Unemployment Compensation System and Labor Market Policies	July 30, 2004
A Performance Effect Analysis of the 5-Year Rehabilitation Project of the Industrial Accident Insurance Scheme	July 30, 2004
The Current Status of Corporate Welfare and Policy Tasks	July 23, 2004
A Study on Improving the Premium Rate Structure of the Industrial Accident Insurance Plan (IAIP)	July 12, 2004
Adjusting the Business Classification for Industrial Accident Insurance and Calculating the Industrial Accident Insurance Rate for 2004	July 9, 2004
Workers' Compensation for Commuting Accidents: A Comparative Study	June 25, 2004
Industrial Accident Insurance for Nonstandard Workers and Special-Type Workers	June 22, 2004
Evaluation of Employer-Provided Training : The Korean Employment Insurance Experience	March 15, 2004
The Employment Insurance System in Canada	August 16, 2003
Assistance Programs for the Unemployed in Australia	June 10, 2003
Rationalization Plans for Pneumoconiosis Policy	May 30, 2003
A Study of Employment Programs in the United States and France	May 30, 2003
A Study on the Balanced Development of Public and Private Pension Systems	May 28, 2003
An International Comparative Study on Social Safety Nets II: The United States	May 15, 2003
Application and Collection System of Industrial Insurance in Major Countries	April 14, 2003

Calculation of By-industry Industrial Accident Insurance Premium Rates in 2003 and the Procedures of and Suggestions from Germany's Industrial Accident Insurance Premium Rate Computation	April 9, 2003
Return-to-Work Assistance System for Injured Workers in Selected Countries	April 9, 2003
Survey of the Ratio of Labor Cost to Output and System Improvement Plans	April 7, 2003
A Comparative Study on Compensation Systems for Work-Related Injuries in Selected Countries	April 7, 2003
A Study on the Fully Funded Method of Financing Industrial Accident Insurance	April 7, 2003
An Analysis on the Status and Determinants of Injured Workers' Return to Work After Industrial Accidents	March 24, 2003
Specialization Plans for Industrial Accident Insurance Facilities	March 20, 2003
The Effects of Unemployment Benefits on the Labor Market	February 28, 2003
Occupational Choices of Women and the Structure of Female Employment	February 28, 2003
International Comparison of Workfare: Europe, the United States, and Korea	December 23, 2002
A Feasibility Study on the Use of the IACI Fund to Finance Industrial Accident Prevention Programs	December 20, 2002
A Feasibility Study on Care Centers for Injured Workers	November 25, 2002
The Employment Insurance System in Japan	November 15, 2002
How to Improve the Public Employment Service in Korea	August 30, 2002
Moral Hazard in the Insurance System: Focusing on Industrial Accident Compensation Insurance	July 25, 2002
Comparative Study on Industrial Accident Compensation Insurance (IACI)	July 18, 2002
The Effectiveness of Measures to Support Employment Maintenance and Methods for Their Improvement	July 5, 2002
Effects of Employment Subsidies for the Disadvantaged and Policy Suggestions	June 20, 2002
Financing the Employment Insurance Fund: Appropriate Level of Premium Rates Under the Possible Scenarios of Institutional Adjustments	June 8, 2002
Implementation of Self-Support Programs and Suggestions for Improvements: Focusing on the Self-Reliance Support Programs of the Ministry of Labor	May 24, 2002
Migrant Workers in Korea	May 15, 2002
A Study on Innovations in the Medical Rehabilitation Fee Schedule of Industrial Accident Compensation Insurance (IACI)	April 30, 2002
Coverage Expansion of Industrial Accident Compensation Insurance (IACI) : Focusing on the Self-Employed and Atypical Workers	April 30, 2002
A Feasibility Study on Care Centers for Injured Workers	April 30, 2002
A Study on Rehabilitation Programs for People with Work-Related Injuries or Illnesses	April 30, 2002
Industrial Accident Compensation Insurance as a Social Safety Net: How to Increase Efficiency of the IACI	April 25, 2002
Policy Initiatives for the Introduction of a Corporate Pension System	April 20, 2002
Changes in Job Stability During the Economic Crisis	April 8, 2002
Reports on Income Inequality and Poverty, and Policy Suggestions	April 3, 2002
Comparative Study on the Social Safety Net [1]: the United Kingdom	February 15, 2002

• Special Topics

KLI Labor Statistics in Selected Countries 2010 2010 KLI Labor Statistics April 27, 2010 March 30, 2010 2009 KLI Labor Statistics August 31, 2009 The 1st-10th Korea Labor and Income Panel Study [KLIPS] Survey Data: Code Book and User's Guide June 30, 2009 The 1st-10th Korea Labor and Income Panel Study [KLIPS] Survey Data: Questionnaire June 30, 2009 10th Wave [2007] Economic Activities of Korean Individuals and Households: Basic Analysis Report of the Korea Labor and Income Panel Study KLI Labor Statistics in Selected Countries 2008 December 31, 2008 LI Labor Statistics in Selected Countries 2008 December 31, 2008 July 23, 2007 July 25, 2007 The 1-8th Korea Labor and Income Panel Study [KLIPS] Survey Data: Code Book and User Guide July 25, 2007 The 1-8th Korea Labor and Income Panel Study [KLIPS] Survey Data: Code Book and User Guide July 23, 2007 July 24, 2007 July 24, 2007 July 28, 2007 July 28, 2007 July 28, 2007 July 28, 2006 July 28,	' '	
The 1st-10th Korea Labor and Income Panel Study (KLIPS) Survey Data: Questionnaire June 30, 2009 The 1st-10th Korea Labor and Income Panel Study (KLIPS) Survey Data: Questionnaire June 30, 2009 10th Wave (2007) Economic Activities of Korean Individuals and Households: Basic Analysis Report of the Korea Labor and Income Panel Study KLI Labor Statistics in Selected Countries 2008 2008 KLI Labor Statistics The 1st-9th Korea Labor and Income Panel Study (KLIPS) Survey Data: Questionnaires, Code Books, and June 30, 2008 1st-9th Korea Labor and Income Panel Study (KLIPS) Survey Data: Questionnaires, Code Books, and June 30, 2008 1st-9th Korea Labor and Income Panel Study (KLIPS) Survey Data: Questionnaires, Code Books, and June 30, 2008 1st-9th Korea Labor and Income Panel: Questionnaires for the 9th Year Survey August 24, 2007 2007 KLI Labor Statistics July 25, 2007 The 1sth Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide June 30, 2007 The 8th 12005) Wave Economic Activities of Korean Households and Individuals: Basic Analysis by Korea Labor and Income Panel Study Effects of Private Tutoring on College Entrance A Study on the Paradigm of Economic and Social Policies in Korea of the Future April 30, 2007 Susues and Challenges of Economic and Social Policies in Korea of the Future April 30, 2007 Composition of Future Korean Social Policies KLI Labor Statistics in Selected Countries 2006 Questionnaires for the 8th Labor Panel Study The 2006 KLI Labor Statistics July 28, 2006 The 7th Study on the Economic Activities of Korean Households and Individuals The 1th Extra Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide Basic Report on the 6th Annual Korean Labor and Income Panel Study May 30, 2005 KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics in Selected Countries 2004 Establishment of a Social Learning Net	KLI Labor Statistics in Selected Countries 2010	April 27, 2010
The 1st-10th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User's Guide The 1st-10th Korea Labor and Income Panel Study (KLIPS) Survey Data: Questionnaire June 30, 2009 MUNE (2007) Economic Activities of Korean Individuals and Households: Basic Analysis Report of the Korea Labor and Income Panel Study KLI Labor Statistics in Selected Countries 2008 2008 KLI Labor Statistics The 1st-9th Korea Labor and Income Panel Study (KLIPS) Survey Data: Questionnaires, Code Books, and User Guides The 9th Wave (2006) Economic Activities of Korean Households and Individuals: Basic Analysis by the Korea Labor and Income Panel Study Korean Labor and Income Panel: Questionnaires for the 9th Year Survey 2007 KLI Labor Statistics The 1st-9th Korea Labor and Income Panel: Questionnaires for the 9th Year Survey 2007 KLI Labor Statistics The 1st-9th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide June 30, 2007 The 8th (2005) Wave Economic Activities of Korean Households and Individuals: Basic Analysis by Korea Labor and Income Panel Study Effects of Private Tutoring on College Entrance A Study on the Panel Guide April 30, 2007 Labor Statistics in Selected Countries 2006 Questionnaires for the 8th Labor Panel Study The 2006 KLI Labor Statistics The 7th Study on the Economic Activities of Korean Households and Individuals The 7th Study on the Economic Activities of Korean Households and Individuals The 7th Study on the Economic Activities of Korean Households and Individuals The 2006 KLI Labor Statistics The 7th Study on the Economic Activities of Korean Households and Individuals The 1-7th Korea Labor and Income Panel Study May 30, 2005 The 1-7th Korea Labor and Income Panel Study May 30, 2005 The 1-7th Korea Labor and Income Panel Study May 30, 2005 April 30, 2004 Establishment of a Social Learning Net	2010 KLI Labor Statistics	March 30, 2010
The 1st-10th Korea Labor and Income Panel Study [KLIPS] Survey Data: Questionnaire 10th Wave (2007) Economic Activities of Korean Individuals and Households: Basic Analysis Report of the Korea Labor and Income Panel Study KLI Labor Statistics in Selected Countries 2008 2008 KLI Labor Statistics The 1st-9th Korea Labor and Income Panel Study [KLIPS] Survey Data: Questionnaires, Code Books, and User Guides The 9th Wave (2008) Economic Activities of Korean Households and Individuals: Basic Analysis by the Korea Labor and Income Panel Study (KLIPS] Survey Data: Questionnaires, Code Books, and User Guides The 9th Wave (2008) Economic Activities of Korean Households and Individuals: Basic Analysis by the Korea Labor and Income Panel: Questionnaires for the 9th Year Survey August 24, 2007 2007 KLI Labor Statistics The 1-8th Korea Labor and Income Panel Study [KLIPS] Survey Data: Code Book and User Guide The 8th [2005] Wave Economic Activities of Korean Households and Individuals: Basic Analysis by Korea Labor and Income Panel Study Effects of Private Tutoring on College Entrance A Study on the Paradigm of Economic and Social Policies in Korea of the Future April 30, 2007 Issues and Challenges of Economic and Social Policies in Korea of the Future Composition of Future Korean Social Policies KLI Labor Statistics in Selected Countries 2006 Questionnaires for the 8th Labor Panel Study The 2006 KLI Labor Statistics The 7th Study on the Economic Activities of Korean Households and Individuals The 7th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide Basic Report on the 6th Annual Korean Labor and Income Panel Study May 30, 2005 KLI Labor Statistics in Selected Countries 2004 April 30, 2004 Establishment of a Social Learning Net April 30, 2004 April 30, 2005	2009 KLI Labor Statistics	August 31, 2009
April 30, 2009 KLI Labor Statistics in Selected Countries 2008 December 31, 2008 KLI Labor Statistics in Selected Countries 2008 December 31, 2008 Zuog KLI Labor Statistics The 1st-9th Korea Labor and Income Panel Study [KLIPS] Survey Data: Questionnaires, Code Books, and User Quides The 9th Wave [2006] Economic Activities of Korean Households and Individuals: Basic Analysis by the Korea Labor and Income Panel Study Korean Labor and Income Panel: Questionnaires for the 9th Year Survey August 24, 2007 Zuog KLI Labor Statistics July 25, 2007 The 1-8th Korea Labor and Income Panel Study [KLIPS] Survey Data: Code Book and User Quide June 30, 2007 The 8th [2005] Wave Economic Activities of Korean Households and Individuals: Basic Analysis by Korea Labor and Income Panel Study Effects of Private Tutoring on College Entrance A Study on the Paradigm of Economic and Social Policies in Korea of the Future April 30, 2007 Composition of Future Korean Social Policies in Korea of the Future April 30, 2007 Composition of Future Korean Social Policies in Korea of the Future April 30, 2007 KLI Labor Statistics in Selected Countries 2006 Questionnaires for the 8th Labor Panel Study The 2006 KLI Labor Statistics July 28, 2006 The 1-7th Korea Labor and Income Panel Study [KLIPS] Survey Data: Code Book and User Quide Basic Report on the 6th Annual Korean Labor and Income Panel Study May 30, 2005 KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics in Selected Countries 2004 April 30, 2005 KLI Labor Statistics in Selected Countries 2004 April 30, 2005 KLI Labor Statistics in Selected Countries 2004 April 30, 2005	The 1st-10th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User's Guide	June 30, 2009
Korea Labor and Income Panel Study KLI Labor Statistics in Selected Countries 2008 2008 KLI Labor Statistics The 1st-9th Korea Labor and Income Panel Study (KLIPS) Survey Data: Questionnaires, Code Books, and User Guides The 9th Wave [2006] Economic Activities of Korean Households and Individuals: Basic Analysis by the Korea Labor and Income Panel Study Korean Labor and Income Panel: Questionnaires for the 9th Year Survey 2007 KLI Labor Statistics The 1-8th Korea Labor and Income Panel: Study (KLIPS) Survey Data: Code Book and User Guide June 30, 2007 The 8th (2005) Wave Economic Activities of Korean Households and Individuals: Basic Analysis by Korea Labor and Income Panel Study Effects of Private Tutoring on College Entrance A Study on the Paradigm of Economic and Social Policies in Korea of the Future April 30, 2007 Issues and Challenges of Economic and Social Policies in Korea of the Future April 30, 2007 KLI Labor Statistics in Selected Countries 2006 Questionnaires for the 8th Labor Panel Study The 2006 KLI Labor Statistics The 7th Study on the Economic Activities of Korean Households and Individuals The 7th Study on the Economic Activities of Korean Households and Individuals The 2006 KLI Labor Statistics The 7th Study on the Economic Activities of Korean Households and Individuals The 2005 KLI Labor Statistics The 7th Study on the Economic Activities of Korean Households and Individuals The 1-7th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide Basic Report on the 6th Annual Korean Labor and Income Panel Study May 30, 2005 KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics 2004 April 30, 2004 April 30, 2005	The 1st-10th Korea Labor and Income Panel Study (KLIPS) Survey Data: Questionnaire	June 30, 2009
The 1st-9th Korea Labor and Income Panel Study (KLIPS) Survey Data: Questionnaires, Code Books, and User Guides The 9th Wave (2006) Economic Activities of Korean Households and Individuals: Basic Analysis by the Korea Labor and Income Panel Study Korean Labor and Income Panel: Questionnaires for the 9th Year Survey August 24, 2007 2007 KLI Labor Statistics July 25, 2007 The 1-8th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide June 30, 2007 The 8th (2005) Wave Economic Activities of Korean Households and Individuals: Basic Analysis by Korea Labor and Income Panel Study Effects of Private Tutoring on College Entrance A Study on the Paradigm of Economic and Social Policies in Korea of the Future April 30, 2007 Issues and Challenges of Economic and Social Policies in Korea of the Future Composition of Future Korean Social Policies KLI Labor Statistics in Selected Countries 2006 Questionnaires for the 8th Labor Panel Study The 2006 KLI Labor Statistics The 7th Study on the Economic Activities of Korean Households and Individuals The 1-7th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide Basic Report on the 6th Annual Korean Labor and Income Panel Study May 30, 2005 KLI Labor Statistics KLI Labor Statistics KLI Labor Statistics KLI Labor Statistics KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics 2004 Establishment of a Social Learning Net		April 30, 2009
The 1st-9th Korea Labor and Income Panel Study [KLIPS] Survey Data: Questionnaires, Code Books, and User Guides The 9th Wave [2006] Economic Activities of Korean Households and Individuals: Basic Analysis by the Korea Labor and Income Panel Study Korean Labor and Income Panel: Questionnaires for the 9th Year Survey 2007 KLI Labor Statistics July 25, 2007 The 1-8th Korea Labor and Income Panel Study [KLIPS] Survey Data: Code Book and User Guide June 30, 2007 The 8th [2005] Wave Economic Activities of Korean Households and Individuals: Basic Analysis by Korea Labor and Income Panel Study Effects of Private Tutoring on College Entrance A Study on the Paradigm of Economic and Social Policies in Korea of the Future April 30, 2007 Labor statistics in Selected Countries 2006 Questionnaires for the 8th Labor Panel Study The 2006 KLI Labor Statistics The 7th Study on the Economic Activities of Korean Households and Individuals The 1-7th Korea Labor and Income Panel Study (KLIPS] Survey Data: Code Book and User Guide Basic Report on the 6th Annual Korean Labor and Income Panel Study May 30, 2005 KLI Labor Statistics April 30, 2005 KLI Labor Statistics in Selected Countries 2004 April 30, 2005	KLI Labor Statistics in Selected Countries 2008	December 31, 2008
The 9th Wave (2006) Economic Activities of Korean Households and Individuals: Basic Analysis by the Korea Labor and Income Panel Study Korean Labor and Income Panel: Questionnaires for the 9th Year Survey August 24, 2007 2007 KLI Labor Statistics July 25, 2007 The 1-8th Korea Labor and Income Panel Study (IKLIPS) Survey Data: Code Book and User Guide June 30, 2007 The 8th [2005] Wave Economic Activities of Korean Households and Individuals: Basic Analysis by Korea Labor and Income Panel Study Effects of Private Tutoring on College Entrance A Study on the Paradigm of Economic and Social Policies in Korea of the Future April 30, 2007 Issues and Challenges of Economic and Social Policies in Korea of the Future April 30, 2007 KLI Labor Statistics in Selected Countries 2006 Questionnaires for the 8th Labor Panel Study The 2006 KLI Labor Statistics July 28, 2006 The 1-7th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide Basic Report on the 6th Annual Korean Labor and Income Panel Study May 30, 2005 KLI Labor Statistics in Selected Countries 2004 April 30, 2004 Establishment of a Social Learning Net	2008 KLI Labor Statistics	July 23, 2008
Labor and Income Panel Study Korean Labor and Income Panel: Questionnaires for the 9th Year Survey 2007 KLI Labor Statistics July 25, 2007 The 1-8th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide June 30, 2007 The 8th (2005) Wave Economic Activities of Korean Households and Individuals: Basic Analysis by Korea Labor and Income Panel Study Effects of Private Tutoring on College Entrance A Study on the Paradigm of Economic and Social Policies in Korea of the Future April 30, 2007 Issues and Challenges of Economic and Social Policies in Korea of the Future April 30, 2007 Composition of Future Korean Social Policies KLI Labor Statistics in Selected Countries 2006 Questionnaires for the 8th Labor Panel Study The 2006 KLI Labor Statistics July 28, 2006 The 1-7th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide Basic Report on the 6th Annual Korean Labor and Income Panel Study May 30, 2005 KLI Labor Statistics KLI Labor Statistics April 30, 2005 KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics 2004 Establishment of a Social Learning Net August 10, 2004 April 30, 2004 Establishment of a Social Learning Net		June 30, 2008
2007 KLI Labor Statistics The 1-8th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide June 30, 2007 The 8th (2005) Wave Economic Activities of Korean Households and Individuals: Basic Analysis by Korea Labor and Income Panel Study Effects of Private Tutoring on College Entrance A Study on the Paradigm of Economic and Social Policies in Korea of the Future April 30, 2007 Issues and Challenges of Economic and Social Policies in Korea of the Future April 30, 2007 Composition of Future Korean Social Policies KLI Labor Statistics in Selected Countries 2006 Questionnaires for the 8th Labor Panel Study The 2006 KLI Labor Statistics July 28, 2006 The 7th Study on the Economic Activities of Korean Households and Individuals The 1-7th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide Basic Report on the 6th Annual Korean Labor and Income Panel Study 2005 KLI Labor Statistics April 30, 2005 KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics 2004 Establishment of a Social Learning Net		June 13, 2008
The 1-8th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide The 8th (2005) Wave Economic Activities of Korean Households and Individuals: Basic Analysis by Korea Labor and Income Panel Study Effects of Private Tutoring on College Entrance A Study on the Paradigm of Economic and Social Policies in Korea of the Future April 30, 2007 Issues and Challenges of Economic and Social Policies in Korea of the Future April 30, 2007 Composition of Future Korean Social Policies April 30, 2007 KLI Labor Statistics in Selected Countries 2006 Questionnaires for the 8th Labor Panel Study The 2006 KLI Labor Statistics July 28, 2006 The 7th Study on the Economic Activities of Korean Households and Individuals The 1-7th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide Basic Report on the 6th Annual Korean Labor and Income Panel Study May 30, 2005 KLI Labor Statistics April 30, 2005 KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics 2004 Establishment of a Social Learning Net	Korean Labor and Income Panel: Questionnaires for the 9th Year Survey	August 24, 2007
The 8th (2005) Wave Economic Activities of Korean Households and Individuals: Basic Analysis by Korea Labor and Income Panel Study Effects of Private Tutoring on College Entrance A Study on the Paradigm of Economic and Social Policies in Korea of the Future April 30, 2007 Issues and Challenges of Economic and Social Policies in Korea of the Future April 30, 2007 Composition of Future Korean Social Policies April 30, 2007 KLI Labor Statistics in Selected Countries 2006 Questionnaires for the 8th Labor Panel Study The 2006 KLI Labor Statistics July 28, 2006 The 7th Study on the Economic Activities of Korean Households and Individuals The 1-7th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide Basic Report on the 6th Annual Korean Labor and Income Panel Study May 30, 2005 KLI Labor Statistics April 30, 2005 KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics 2004 Establishment of a Social Learning Net	2007 KLI Labor Statistics	July 25, 2007
Effects of Private Tutoring on College Entrance A Study on the Paradigm of Economic and Social Policies in Korea of the Future April 30, 2007 Issues and Challenges of Economic and Social Policies in Korea of the Future April 30, 2007 Composition of Future Korean Social Policies April 30, 2007 KLI Labor Statistics in Selected Countries 2006 Questionnaires for the 8th Labor Panel Study The 2006 KLI Labor Statistics July 28, 2006 The 7th Study on the Economic Activities of Korean Households and Individuals June 29, 2006 The 1-7th Korea Labor and Income Panel Study [KLIPS] Survey Data: Code Book and User Guide Basic Report on the 6th Annual Korean Labor and Income Panel Study May 30, 2005 KLI Labor Statistics April 30, 2005 KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics 2004 Establishment of a Social Learning Net	The 1-8th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide	June 30, 2007
A Study on the Paradigm of Economic and Social Policies in Korea of the Future April 30, 2007 Issues and Challenges of Economic and Social Policies in Korea of the Future April 30, 2007 Composition of Future Korean Social Policies April 30, 2007 KLI Labor Statistics in Selected Countries 2006 Questionnaires for the 8th Labor Panel Study The 2006 KLI Labor Statistics July 28, 2006 The 7th Study on the Economic Activities of Korean Households and Individuals June 29, 2006 The 1-7th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide Basic Report on the 6th Annual Korean Labor and Income Panel Study May 30, 2005 KLI Labor Statistics April 30, 2005 KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics 2004 Establishment of a Social Learning Net		June 30, 2007
Issues and Challenges of Economic and Social Policies in Korea of the Future April 30, 2007 Composition of Future Korean Social Policies April 30, 2007 KLI Labor Statistics in Selected Countries 2006 Questionnaires for the 8th Labor Panel Study The 2006 KLI Labor Statistics July 28, 2006 The 7th Study on the Economic Activities of Korean Households and Individuals June 29, 2006 The 1-7th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide Basic Report on the 6th Annual Korean Labor and Income Panel Study May 30, 2005 KLI Labor Statistics April 30, 2005 KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics 2004 Establishment of a Social Learning Net	Effects of Private Tutoring on College Entrance	May 16, 2007
Composition of Future Korean Social Policies April 30, 2007 KLI Labor Statistics in Selected Countries 2006 Questionnaires for the 8th Labor Panel Study The 2006 KLI Labor Statistics July 28, 2006 The 7th Study on the Economic Activities of Korean Households and Individuals The 1-7th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide Basic Report on the 6th Annual Korean Labor and Income Panel Study May 30, 2005 KLI Labor Statistics April 30, 2005 KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics 2004 Establishment of a Social Learning Net	A Study on the Paradigm of Economic and Social Policies in Korea of the Future	April 30, 2007
KLI Labor Statistics in Selected Countries 2006 Questionnaires for the 8th Labor Panel Study The 2006 KLI Labor Statistics July 28, 2006 The 7th Study on the Economic Activities of Korean Households and Individuals June 29, 2006 The 1-7th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide Basic Report on the 6th Annual Korean Labor and Income Panel Study May 30, 2005 KLI Labor Statistics April 30, 2005 KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics 2004 Establishment of a Social Learning Net	Issues and Challenges of Economic and Social Policies in Korea of the Future	April 30, 2007
Questionnaires for the 8th Labor Panel Study The 2006 KLI Labor Statistics July 28, 2006 The 7th Study on the Economic Activities of Korean Households and Individuals June 29, 2006 The 1-7th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide Basic Report on the 6th Annual Korean Labor and Income Panel Study May 30, 2005 2005 KLI Labor Statistics April 30, 2005 KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics 2004 Establishment of a Social Learning Net August 10, 2006 July 28, 2006 June 29, 2006 June 20, 2006 April 30, 2005 April 30, 2005 April 30, 2004 April 29, 2003	Composition of Future Korean Social Policies	April 30, 2007
The 2006 KLI Labor Statistics July 28, 2006 The 7th Study on the Economic Activities of Korean Households and Individuals June 29, 2006 The 1-7th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide Basic Report on the 6th Annual Korean Labor and Income Panel Study May 30, 2005 2005 KLI Labor Statistics April 30, 2005 KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics 2004 Establishment of a Social Learning Net April 29, 2003	KLI Labor Statistics in Selected Countries 2006	December 30, 2006
The 7th Study on the Economic Activities of Korean Households and Individuals The 1-7th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide June 29, 2006 Basic Report on the 6th Annual Korean Labor and Income Panel Study May 30, 2005 2005 KLI Labor Statistics April 30, 2005 KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics 2004 April 30, 2004 Establishment of a Social Learning Net	Questionnaires for the 8th Labor Panel Study	August 10, 2006
The 1-7th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide Basic Report on the 6th Annual Korean Labor and Income Panel Study May 30, 2005 2005 KLI Labor Statistics KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics 2004 KLI Labor Statistics 2004 Establishment of a Social Learning Net June 20, 2006 May 30, 2005 April 30, 2005 April 30, 2004 April 30, 2004	The 2006 KLI Labor Statistics	July 28, 2006
Basic Report on the 6th Annual Korean Labor and Income Panel Study 2005 KLI Labor Statistics KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics 2004 KLI Labor Statistics 2004 Establishment of a Social Learning Net May 30, 2005 April 30, 2005 April 30, 2004 April 29, 2003	The 7th Study on the Economic Activities of Korean Households and Individuals	June 29, 2006
2005 KLI Labor Statistics KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics 2004 KLI Labor Statistics 2004 Establishment of a Social Learning Net April 30, 2005 April 30, 2004 April 29, 2003	The 1-7th Korea Labor and Income Panel Study (KLIPS) Survey Data: Code Book and User Guide	June 20, 2006
KLI Labor Statistics in Selected Countries 2004 KLI Labor Statistics 2004 April 30, 2004 Establishment of a Social Learning Net April 29, 2003	Basic Report on the 6th Annual Korean Labor and Income Panel Study	May 30, 2005
KLI Labor Statistics 2004 April 30, 2004 Establishment of a Social Learning Net April 29, 2003	2005 KLI Labor Statistics	April 30, 2005
Establishment of a Social Learning Net April 29, 2003	KLI Labor Statistics in Selected Countries 2004	December 30, 2004
	KLI Labor Statistics 2004	April 30, 2004
Methods to Improve Labor Statistics April 25, 2003	Establishment of a Social Learning Net	April 29, 2003
	Methods to Improve Labor Statistics	April 25, 2003

KLI Labor Statistics in Selected Countries 2002	January 2, 2003
The Fourth Annual Report of the KLIPS: Economic Activities of Households and Individuals in Korea (IV)	December 20, 2002
A Study on Commitment in M&A Firms	December 13, 2002
ICT and Skilled Labor	November 29, 2002
Labor Reforms in Socialist Countries and a Model for North Korea	June 15, 2002
Investment in Vocational Education and Training in Company, and its Challenges	May 30, 2002
KLI Labor Statistics 2002	April 30, 2002
The Third Annual Report of the KLIPS	April 25, 2002

Chronology

May 1988	Mooki Bai appointed the 1st President
Dec. 1988	Promulgation of the Law establishing the KLI (Law No.4078)
Sep. 1989	Establishment of the Advanced Labor-Management Relations Program
Dec. 1995	Establishment of the Employment Research Center
Feb. 1999	Establishment of the Research Center for Industrial Safety and Welfare
Mar. 2004	Establishment of the Labor Insurance Research Center which combined the Employment Insurance Research Center and the Research Center for Industrial Safety and Welfare
Mar. 2004	Establishment of the New Paradigm Center as an affiliated body
Jan. 2005	Establishment of the Data Center
Feb. 2005	Establishment of the Center for Wage and Job Research
Jun. 2006	Ranked first in annual evaluation among the 23 Research Institutes under the National Research Council for Economics, Humanities, and Social Sciences
Jun. 2007	Ranked second in annual evaluation among the 23 Research Institutes under the National Research Council for Economics, Humanities, and Social Sciences
Jan. 2009	New Paradigm Center changed to Korea Workplace Innovation Center